

W A M Capital

Level 26, Governor Phillip Tower 1 Farrer Place Sydney NSW 2000 Australia

13 October 2020

BY ELECTRONIC LODGEMENT

Company Announcements ASX Limited Exchange Centre 20 Bridge Street Sydney NSW 2000

Off-Market Takeover Bid for Contango Income Generator Limited

In accordance with section 633(1) item 8 of the Corporations Act, WAM Capital Limited (ASX: WAM) (WAM) gives notice that it has today completed the dispatch of its bidder's statement and offers (Bidder's Statement) in relation to its off-market takeover bid for shares in Contango Income Generator Limited (ASX: CIE) (CIE).

The offers are dated 13 October 2020.

A copy of the Bidder's Statement as sent to CIE's shareholders is attached and has been lodged with the Australian Securities and Investments Commission and sent to CIE earlier today.

WAM's offer is now open for acceptance and WAM encourages CIE shareholders to accept the offer as soon as possible. If the offer is successful, CIE Shareholders accepting the offer may either:

- a) remain a WAM Shareholder; or
- b) utilise WAM's superior on-market liquidity to exit their position,

for further information, please refer to the Bidder's Statement.

WAM's offer is scheduled to close on 13 November 2020, unless extended or withdrawn.

Yours faithfully,

Geoff Wilson AO Chairman



This is an important document and requires your immediate attention. If you are in any doubt as to how to deal with it, you should consult your financial or other professional advisor.

W A M Capital

Bidder's Statement

Offer to acquire all the ordinary shares in Contango Income Generator Limited (ASX: CIE) (ACN 160 959 991)

Ву

WAM Capital Limited (ASX: WAM) (ACN 086 587 395)

Offer of

1 WAM Share for every 3 of your CIE Shares

To accept the Offer please complete the enclosed Acceptance Form and use the enclosed reply-paid envelope to return the completed Acceptance Form to us or contact your Broker. The Offer is dated 13 October 2020 and will expire at 7.00pm (Sydney time) on 13 November 2020 (unless withdrawn or extended). ASIC takes no responsibility for this Bidder's Statement.

Important Information

Key dates

Event	Date
Date of Bidder's Statement	28 September 2020
Dispatch of Bidder's Statement and Offer opens (date of Offer)	13 October 2020
Date of Offer Conditions notice	6 November 2020
Close of Offer*	13 November 2020

^{*}This date is indicative only and may be changed as permitted by the Corporations Act.

Bidder's Statement and Offer

This is a Bidder's Statement dated 28 September 2020 given by WAM to CIE and each holder of CIE Shares under Part 6.5 of Chapter 6 of the Corporations Act. This Bidder's Statement was approved by a unanimous resolution of the Board of WAM. It includes an Offer for CIE Shares dated 13 October 2020. The Offer extends to CIE Shares on issue on 30 September 2020 and any CIE Shares issued during the Offer Period.

ASIC and ASX

A copy of this Bidder's Statement was lodged with ASIC and the ASX on 28 September 2020. The ASX, ASIC and their officers take no responsibility for the contents of this Bidder's Statement.

Investment decision

This is an important document. This Bidder's Statement does not take into account the individual investment objectives, financial situation or any particular needs of each CIE Shareholder or any other person. CIE Shareholders should seek independent financial and taxation advice before deciding whether or not to accept the Offer.

Disclaimer

Except for historical information contained in this booklet, there may be matters discussed in this Bidder's Statement that are forward-looking statements. Such statements are only predictions and are subject to inherent risks and uncertainty. Forward-looking statements, which are based on assumptions and estimates and describe WAM's future plans, strategies and expectations, are generally identifiable by the use of the words 'anticipate', 'will', 'believe', 'estimate', 'expect', 'intend', 'seek' or similar expressions. CIE Shareholders are cautioned not to place undue reliance on forward-looking statements. By its nature, forward-looking information involves numerous assumptions, inherent risks and uncertainties both general and specific that contribute to the possibility that the predictions, forecasts, projections and other forward-looking statements will not occur. Those risks and uncertainties include factors and risks specific to the industry in which WAM and CIE operate as well as general economic conditions and

prevailing exchange rates and interest rates. Actual events or results may differ materially.

All forward-looking statements attributable to WAM, or persons acting on behalf of WAM, are expressly qualified in their entirety by the cautionary statements in this Section. Except as expressly required by the Corporations Act, WAM makes no undertaking to publicly update or revise any forward-looking statements provided in this Bidder's Statement, whether as a result of new information, future events or otherwise, or the list of risks affecting this information.

None of WAM, WAM's Directors, any person named in this Bidder's Statement or any other person involved in the preparation of this Bidder's Statement makes any representation or warranty (express or implied) as to the accuracy or likelihood of fulfilment of any forward-looking statement, except to the extent required by law.

Distribution outside Australia

The distribution of this Bidder's Statement may, in some countries, be restricted by law or regulation. Accordingly, persons who come into possession of this Bidder's Statement should inform themselves of, and observe, those restrictions.

Important notice to US shareholders

This Bidder's Statement does not constitute an offer to sell, or a solicitation of an offer to buy, any securities in the United States (US). The securities described herein have not been, and will not be, registered under the US Securities Act of 1933, as amended (US Securities Act) or the securities laws of any state or other jurisdiction of the United States. Accordingly, such securities may not be offered or sold, directly or indirectly, to any person in the United States except pursuant to an exemption from, or in a transaction not subject to, the registration requirements of the US Securities Act and applicable US state securities laws. This Bidder's Statement may not be forwarded, distributed or reproduced in whole or in part in any jurisdiction where it would be unlawful, including the United States.

The Offer is made for the shares of an Australian registered company. It is important that US shareholders understand that the Offer is subject to disclosure requirements in Australia that are materially different from those in the US. US shareholders should be aware that, subject to the Corporations Act, WAM may purchase CIE Shares otherwise than under the Offer, such as in open market during or after the Offer Period or privately negotiated purchases after the end of the Offer Period.

Notice to all other Foreign Shareholders outside the US

The Offer and this Bidder's Statement are subject to Australian disclosure requirements which may be different from those applicable in other jurisdictions. This Bidder's Statement and the Offer do not constitute an offer in any place which, or to any person to whom, it would not be lawful to make such an offer.

The distribution of this Bidder's Statement may, in some countries, be restricted by law or regulation. Accordingly, persons who come into possession of this Bidder's Statement should inform themselves of, and observe, those restrictions.

Defined terms

Defined terms are capitalised throughout this Bidder's Statement. Definitions of these terms are set out in Section 12. Unless the contrary intention appears, the context requires otherwise, or words are defined in Section 12, words and phrases in this Bidder's Statement have the same meaning and interpretation as in the Corporations Act.

Sources of information

Information included in this Bidder's Statement relating to CIE and its business has been derived solely from publicly available sources published by CIE, including its annual reports and CIE's announcements to the ASX and on its website.

Subject to the foregoing and to the maximum extent permitted by law, WAM, WAM's Directors, each person named in this Bidder's Statement and each other person involved in the preparation of this Bidder's Statement disclaim all liability for any information concerning CIE included in this Bidder's Statement. CIE Shareholders should form their own views concerning CIE from publicly available information.

Privacy statement

WAM has collected information from the register of CIE Shareholders. The Corporations Act permits information to be made available to certain persons, including WAM. Your information may also be disclosed on a confidential basis to WAM's related bodies corporate and external service providers and may be required to be disclosed to regulators such as ASIC. You can contact us for details of information held by us about you.

Directory

Bidder

WAM Capital Limited Level 26, Governor Phillip Tower 1 Farrer Place Sydney NSW 2000

T: (02) 9247 6755

E: <u>info@wilsonassetmanagement.com.au</u>
W: <u>wilsonassetmanagement.com.au</u>

Investment Manager of WAM

Wilson Asset Management (International) Pty Limited Level 26, Governor Phillip Tower 1 Farrer Place Sydney NSW 2000

Share Registry

Boardroom Pty Limited Level 12, 225 George Street Sydney NSW 2000 T: (02) 9290 9600

F: (02) 9279 0664

WAM Board

Geoff Wilson AO (Chairman) Kate Thorley Dr Philippa Ryan James Chirnside Lindsay Mann Matthew Pancino

Company Secretary

Linda Kiriczenko

Australian Legal Advisor

Principal Legal Adviser
Mills Oakley
Level 7, 151 Clarence Street
Sydney NSW 2000
T: (02) 8289 5800

W: www.millsoakley.com.au

Original Legal Adviser
PwC Australia
One International Towers
Watermans Quay
Barangaroo NSW 2000
T: (02) 8266 0000

W: www.pwc.com.au/legal

Contents

Impor	tant Information	1
Direct	ory	3
Conte	nts	4
Chairr	man's letter	5
Summ	nary of the Offer	9
The O	ffer provides two significant benefits to all CIE Shareholders	11
1.	Key Questions	15
2.	About WAM Capital Limited	21
3.	Information about WAM Shares	28
4.	About Contango Income Generator Limited	30
5.	Information about CIE Shares	37
6.	Profile of the Enlarged Group	38
7.	WAM's intentions	47
8.	Risk factors	51
9.	Sources of Consideration	58
10.	Tax considerations	59
11.	Additional information	63
12.	Definitions and interpretations	72
13.	Approval of Bidder's Statement	76
Sched	ule 1 – Terms of the Offer	77
1.	Offer	77
2.	Offer Consideration	77
3.	Foreign Shareholders	77
4.	Fractional entitlements	77
5.	Offer Period	78
6.	How to accept this Offer	78
7.	Provision of consideration	80
8.	Conditions of the Offer	82
9.	Acceptances by transferees and nominees	88
10.	Other matters	89
11.	Costs and expenses	. 90
12.	Official Quotation of WAM Shares	90
Sched	lule 2 – WAM Announcements since 30 June 2020	91

Chairman's letter

W A M Capital



Dear Fellow Shareholders,

Contango Income Generator Limited (ASX: CIE) has delivered deeply disappointing results and failed to provide shareholder value. We present WAM Capital Limited's Offer to acquire up to 100% of CIE Shares on highly beneficial terms that provides a meaningful choice about the future of your investment: remain a WAM shareholder or utilise WAM's superior onmarket liquidity to exit your position.

This Offer allows CIE Shareholders to exit their CIE Shares by acquiring 1 WAM Share for every 3 CIE Shares held. This represents a premium of 17.6% to CIE's Share price on 25 September 2020, 18.1% based on CIE's one-month volume-weighted average price (VWAP) and a premium to CIE's pre-tax net tangible assets (NTA) as at 31 August 2020. CIE Shareholders have suffered a persistent and deepening share price discount to NTA, poor investment portfolio performance, an illogical change in investment strategy and an increase in fees, dilution of shareholder value, poor corporate governance and the CIE Board's failure to present Wilson Asset Management's superior alternative proposal.

Acquiring WAM shares provides access to Wilson Asset Management's investment expertise and experience and commitment to shareholder engagement, and WAM's track record of investment portfolio outperformance, greater market capitalisation and on-market liquidity, and lower management expense ratio.

The case for change

A persistent and deepening share price discount to NTA, an illogical change in investment strategy and an increase in fees

CIE shares have traded at a persistent discount to their underlying pre-tax NTA throughout its history, with an average share price discount to CIE's pre-tax net tangible assets of 13.2% over the past 12 months. The CIE Board's proposal to substantially change the investment strategy of CIE to a global long short strategy and to appoint a US investment manager with a poor track record in managing an LIC was flawed and on unfavourable terms, serving only to prolong the pain of CIE Shareholders. Shares in the single listed investment company (LIC) managed by WCM Investment Management currently trade at a 10.8%¹ discount to their NTA backing, and yet the CIE Board's proposal to enter into the new CIE Investment Management Agreement has seen an increase in the management fees paid by CIE Shareholders, from 0.95% per annum (with no performance fee) to 1.4% plus a 20% performance fee, as well as an early termination fee of up to \$500,000.

¹ Based on the WCM Global Growth Limited (ASX: WQG) share price of \$1.31 as at 25 September 2020 and the 18 September 2020 reported pre-tax NTA of \$1.469 per share.



Dilution of value and poor corporate governance

On 12 August 2020, the CIE Board announced it had completed a Placement to professional and sophisticated investors at a 14.9% discount to CIE's 31 July 2020 pre-tax NTA. The CIE Board also announced that it intended to conduct a share purchase plan (SPP) for existing CIE Shareholders.

The CIE Board raised new capital that diluted 88% of CIE Shareholders, with only 269 out of a register of 2,295 participating in the recent SPP. The CIE Board has effectively taken over \$2 million of value from CIE Shareholders and given it to participants in the Placement and SPP.

The failure to present Wilson Asset Management's superior alternative proposal

Wilson Asset Management submitted a proposal to the CIE Board on 13 August 2020 to reposition CIE as an active investor in the highest quality Australian companies, delivering investors a stream of fully franked dividends while providing capital growth and preserving shareholder capital, managed by Wilson Asset Management (**Proposal**). The CIE Board refused to announce this Proposal to CIE Shareholders. The Proposal was subsequently announced to the WAM ASX announcements platform on 18 August 2020 but has still not been communicated to CIE Shareholders by the CIE Board.

The Proposal to welcome CIE Shareholders into the stable of Wilson Asset Management LICs would provide CIE Shareholders with access to a proven investment manager and LIC specialist with a 21-year track record, under more favourable terms than the WCM Investment Management proposal. We are disappointed that the CIE Board failed its shareholders by not putting the Wilson Asset Management Proposal to be voted on by CIE Shareholders at the recent CIE extraordinary general meeting (EGM) on 18 September 2020. Before the EGM, we requested the shareholder register of CIE from the CIE Board on 18 August 2020 in order to inform CIE shareholders of the Wilson Asset Management Proposal, following the CIE Board's failure to do so. The CIE Board breached the Corporations Act by failing to provide the shareholder register in the required timeframe. After we threatened to commence Court proceedings to compel CIE to comply with the request for the share register, the CIE Board only then sent us the requested register.

The CIE Board has actively denied CIE Shareholders the opportunity to vote on the Wilson Asset Management superior Proposal at the recent EGM. At the CIE EGM on 18 September, 47% of shareholders voted against the CIE Board's resolution to adopt a new investment strategy with WCM Investment Management.

On 11 September 2020, WAM and its associated entities (**WAM Entities**) lodged a notice under section 249D of the Corporations Act, requesting that CIE call and arrange an extraordinary general meeting of shareholders (**WAM EGM**) at which WAM would seek to put a resolution to members to remove Mark Kerr and Don Clarke as CIE Directors, and appoint myself and Glenn Burge in their place, with a view to subsequently appointing Wilson Asset Management as CIE's new investment manager. The CIE Board failed to advise CIE Shareholders of our section 249D notice in the required timeframe of the ASX Listing Rules and CIE has not yet called the WAM EGM. WAM is proceeding with this Offer regardless. See Section 7 for more detail on WAM's intentions in relation to CIE.

An exit opportunity with WAM Capital

In making this Offer, WAM is providing all CIE Shareholders with the opportunity to exit their positions in CIE at an 18.1% premium to CIE's one-month VWAP of \$0.632.²

We consider that the recent actions by the CIE Board represent a pattern of poor decision-making on behalf of CIE Shareholders. The CIE Board have been unable to close the persistent CIE Share price discount to NTA and employed inconsistent and ineffective capital management initiatives to date.

If this Offer is successful, WAM will provide CIE Shareholders with a choice, either:

- a) remain a WAM Shareholder; or
- b) utilise WAM's superior on-market liquidity to exit their position.

For further information see pages 11 to 14 of this Bidder's Statement.

If you accept the Offer and are issued WAM Shares under the Offer before the WAM Dividend record date on 20 November 2020, and you continue to hold your WAM Shares on that date, you will also receive the fully franked WAM dividend of 7.75 cents per WAM Share announced on 8 July 2020 and payable on 27 November 2020 (**WAM Dividend**).

If the Offer is extended, and does not become unconditional, you may not receive your WAM Shares in time to receive the WAM Dividend. If the Offer does not become unconditional and is extended for less than one month, you will not be able to withdraw your acceptance of the Offer, and you will not receive the WAM Dividend.

In order to receive the WAM Dividend, it is important that you accept the Offer as early as possible and in any event on or before 13 November 2020 to receive your WAM Shares by 20 November 2020.

CIE Shareholders are being offered consideration under the Offer that comprises a specified number of WAM Shares, rather than a number of WAM Shares with a specified market value. As a result, the value of the consideration will fluctuate depending on the value of WAM Shares.

-

² The premium calculations are based on an implied WAM Share price of \$2.24 as at 25 September 2020 and the underlying VWAP information has been sourced for CIE Shares using IRESS.

Conditions of the Offer

The Offer is subject only to the following Defeating Conditions as summarised below – refer to Schedule 1 clause 8 for more detail:

- a) before the end of the Offer Period, WAM has a Relevant Interest in at least 50.1% (by number) of CIE Shares;
- b) between the date of this Bidder's Statement and the end of the Offer Period the Investment Management Agreement is not varied or assigned and no new investment management agreement is entered into (or CIE enters into an agreement in relation to any of the foregoing) other than assignment to, or entry into an agreement with, Wilson Asset Management;
- c) between the date of this Bidder's Statement and the end of the Offer Period no Prescribed Occurrence occurs;
- d) the pre-tax NTA of CIE does not decline by 7.5% or more below the pre-tax net NTA of CIE of \$0.746 per share, announced to the ASX in the Monthly NTA Statement 31 August 2020 on 4 September 2020;
- e) the Dow Jones Industrial Average, in AUD terms, does not close at a level that is 5% or more below the level of that index at 5.00pm (Sydney time) on the trading day immediately prior to the Announcement Date and remaining at or below that level for at least two consecutive trading days;
- between the date of this Bidder's Statement and the end of the Offer Period there is no material adverse change to the assets, liabilities, financial position, performance, profitability or prospects of CIE and/or its Subsidiaries;
- g) between the date of this Bidder's Statement and the end of the Offer Period CIE and/or its Subsidiaries do not undertake certain actions such as making or proposing to make any variations to its constitution, giving or agreeing to give any encumbrance over any of its assets, or appointing any additional Director to its Board of Directors; and
- h) between the date of this Bidder's Statement and the end of the Offer Period there is no regulatory action which materially affects the Offer.

Please see Schedule 1 of this Bidder's Statement for further details of the Conditions to the Offer. The Offer is scheduled to close at 7:00pm (AEST) on 13 November 2020.

I encourage you to read this Bidder's Statement for more details about the Offer and about WAM. If you have any questions, please call me on 02 9247 6755 or our Chief Financial Officer Jesse Hamilton on 0401 944 807. For inquiries related to your personal financial situation please contact your legal, financial or other professional adviser.

Yours faithfully,

Geoff Wilson AO Chairman

Summary of the Offer

The information set out below is only a summary of the Offer. You should read the entire Bidder's Statement before deciding whether to accept the Offer.

The Offer	WAM is offering to acquire all of your CIE Shares. See Section 2 for information relating to WAM.
Offer Consideration	You are being offered 1 WAM Share for every 3 CIE Shares you own.
Closing date	The Offer is scheduled to close at 7:00pm (Sydney time) on 13 November 2020 (but it may be extended).
Offer Conditions	The Offer is subject to the conditions (Defeating Conditions) detailed at clause 8.1 of Schedule 1 of this Bidder's Statement, and also summarised in the Chairman's letter included in this Bidder's Statement, and Section 1.7.
Settlement terms	If you accept this Offer, WAM will, in the usual course, issue the WAM Shares to which you are entitled on or before the earlier of:
	 the day one month after you accept this Offer or, if the Offer is subject to Conditions when accepted, one month after the contract resulting from your acceptance becomes unconditional; and
	b) the day 21 days after the end of the Offer Period.
	Full details of when you will be issued the consideration are set out in Schedule 1. It is WAM's intention, however, to issue the Offer Consideration within seven days of the end of the Offer Period (provided that the Offer is then unconditional).
How to accept	CHESS Holdings
the Offer	If your CIE Shares are held in a CHESS Holding, to accept the Offer you must either:
	 instruct your controlling participant (usually your Broker) to accept the Offer on your behalf; or
	b) complete, sign and return the enclosed Acceptance Form in accordance with the instructions contained within it.
	Issuer Sponsored Holdings
	If your CIE Shares are held through an Issuer Sponsored Holding, to accept the Offer you must complete, sign and return Acceptance Form in accordance with the instructions contained within it. Signed Acceptance Forms must be received before the end of the Offer Period at:
	By mail: Contango Income Generator Limited Takeover C/- Boardroom Pty Limited GPO Box 3993 Sydney NSW 2001
	Hand delivered: Boardroom Pty Limited Level 12, 225 George Street Sydney NSW 2000
	A reply-paid envelope is enclosed. If you wish to sell your CIE Shares on-market, please contact your Broker.

No stamp duty You will not pay any stamp duty on accepting the Offer. If you have an Issuer Sponsored Holding on acceptance of the Offer you your CIE Shares for WAM Shares without incurring brokerage fees. If your CIE Shares are registered in a CHESS Holding, you will generally brokerage fees in accepting the Offer (although you should ask your Offer). If you sell your CIE Shares on the ASX instead of accepting the Offer, you brokerage fees (and potentially GST on those fees).	y not incur any Controlling				
your CIE Shares for WAM Shares without incurring brokerage fees. If your CIE Shares are registered in a CHESS Holding, you will generally brokerage fees in accepting the Offer (although you should ask your CP Participant, normally your broker, to confirm). If you sell your CIE Shares on the ASX instead of accepting the Offer, you	y not incur any Controlling				
brokerage fees in accepting the Offer (although you should ask your Offer (although you should ask your Offer, to confirm). If you sell your CIE Shares on the ASX instead of accepting the Offer, you	Controlling				
	ou may incur				
brokerage rees (una potentiary as ron those rees).					
you will accept the Offer for your CIE Shares.	You should read this Bidder's Statement in its entirety before deciding whether or not you will accept the Offer for your CIE Shares. If you wish to accept the Offer, please see Schedule 1 of this Bidder's Statement.				
Further information For questions about your CIE Shares, the Offer, or how to accept the 0 to the remainder of this Bidder's Statement. If you have any question Chairman Geoff Wilson AO on 02 9247 6755 or Chief Financial Officer on 0401 944 807.	s please call				
Offers to Foreign	that:				
Shareholders a) it is lawful and not unduly onerous and not unduly impracticable to you and to issue you with WAM Shares on acceptance of the C					
b) it is not unlawful for you to accept the Offer by the law of the release outside Australia and its external territories,	evant place				
you will not be entitled to receive WAM Shares as consideration for y	our CIE Shares.				
If you are a Foreign Shareholder and you accept the Offer, or your CIE compulsorily acquired, you will not receive WAM Shares. Instead you cash amount calculated under clause 7.3 of Schedule 1 as part of the process explained in Section 11.12.	will receive the				
Further information about the entitlement of a Foreign Shareholder is Section 11.12.	s set out in				

The above information is a summary only of key elements of this Bidder's Statement. CIE Shareholders should read this Bidder's Statement in its entirety before making any decision in relation to the Offer.

The Offer provides two significant benefits to all CIE Shareholders

1. A highly attractive premium³

The Offer Consideration is 1 WAM Share for every 3 CIE Shares you own. This represents a highly attractive premium over the historical market price of CIE Shares.

The Offer is equivalent to \$0.747 per CIE Share, a 17.6% premium to the closing price of CIE Shares the trading day before the date of this Bidder's Statement of \$0.635 and represents:

- a 18.1% premium to the VWAP of CIE Shares for one month before the date of this Bidder's Statement;
- a 17.6% premium to the VWAP of CIE Shares for three months before the date of this Bidder's Statement; and
- a 23.6% premium to the VWAP of CIE Shares for six months before the date of this Bidder's Statement.

CIE Shares are currently trading at a 16.2% discount to CIE's pre-tax NTA as at 31 August 2020 and were at:

- a 12.1% CIE Share price discount to the pre-tax NTA as at 31 July 2020;
- a 20.2% CIE Share price discount to the pre-tax NTA as at 31 May 2020; and
- a 12.7% CIE Share price discount to the pre-tax NTA as at 31 March 2020.

While there are many factors that may influence the market price of CIE Shares, there is a risk that CIE Shares may fall after the close of the Offer.

In contrast with CIE Shares, at the Announcement Date WAM Shares are trading at a 29.1% premium to WAM's pre-tax NTA and were at:

- a 19.9% WAM Share price premium to the pre-tax NTA as at 31 July 2020;
- a 22.8% WAM Share price premium to the pre-tax NTA as at 31 May 2020; and
- a 23.4% WAM Share price premium to the pre-tax NTA as at 31 March 2020.

2. A choice about their future as a shareholder

Subject to the Offer being successful WAM is providing CIE Shareholders who accept the Offer with a choice, either:

- a) remain a WAM Shareholder; or
- b) utilise WAM's superior on-market liquidity to exit their position.

W A M Capital

³ The premium calculations are based on an implied WAM Share price of \$2.24 as at 25 September 2020 and the underlying VWAP information has been sourced for CIE Shares using IRESS.

a) Remain a WAM shareholder

The benefits of taking up the Offer and remaining a WAM Shareholder include gaining access to:

- Wilson Asset Management's investment expertise and experience;
- Wilson Asset Management's commitment to shareholder engagement;
- WAM's track record of investment portfolio outperformance;
- WAM's greater market capitalisation and on-market liquidity; and
- WAM's lower management expense ratio.

Wilson Asset Management's investment expertise and experience

Established in 1997 by Geoff Wilson AO, the investment manager of WAM, Wilson Asset Management is an independently owned investment manager and LIC specialist based in Sydney, Australia, responsible for investing more than \$3.5 billion in Australian and international companies on behalf of 88,000 retail investors across seven LICs. The investment team comprises 14 highly experienced professionals with more than 190 years' combined investment experience and a total focus on managing money. Wilson Asset Management has a strong track record of delivering risk-adjusted returns for shareholders and making a difference for investors and the community for more than 20 years.

Wilson Asset Management created and is the lead supporter of the first LICs to deliver both investment and social returns: Future Generation Australia (ASX: FGX) and Future Generation Global (ASX: FGG). Wilson Asset Management advocates and acts for retail investors, is a member of the global philanthropic Pledge 1% movement and provides all team members with \$10,000 each year to donate to charities of their choice. This philanthropic investment is made by Wilson Asset Management.

For further information about WAM see Section 2.

Wilson Asset Management's commitment to shareholder engagement

Shareholders are the owners of a LIC. Wilson Asset Management believes shareholder engagement is crucial to the success of a LIC and provides:

- Updates from our Lead Portfolio Managers;
- Investment team insights at WAM Vault: wilsonassetmanagement.com.au/vault;
- Regular roundtables with our shareholders and planners, advisers and their clients;
- NTA reports and investment updates;
- Shareholder Q&A calls;
- Presentations and lunches across Australia; and
- Annual and interim results announcements.

WAM's track record of investment portfolio outperformance

WAM's investment portfolio has returned 16.1% per annum over more than 20 years, outperforming the S&P/ASX All Ordinaries Accumulation Index by 8.1%.

Performance at 31 August 2020	FYTD	1 yr	3 yrs %pa	5 yrs %pa		Since inception % pa (Aug-99)
WAM Capital Investment Portfolio	11.7%	4.4%	7.7%	10.8%	13.1%	16.1%
S&P/ASX All Ordinaries Accumulation Index	4.7%	-3.5%	6.7%	7.9%	7.9%	8.0%
Outperformance	+7.0%	+7.9%	+1.0%	+2.9%	+5.2%	+8.1%

Investment performance and Index returns are before expenses, fees and taxes.

WAM's greater market capitalisation and on-market liquidity

WAM believes that shares in larger LICs are more likely to trade at a premium to their pretax NTA than the shares of smaller LICs. WAM has a market capitalisation of \$1,625.0 million as at 25 September 2020. Assuming the Offer is successful and WAM acquires 90% of CIE Shares and proceeds to compulsory acquisition, the combination of CIE and WAM will be significantly larger than CIE on a standalone basis. Based on the pre-tax NTA of WAM and CIE as at 31 August 2020, the Enlarged Group would have a combined pre-tax NTA of approximately \$1,339.2 million, which would make it one of the largest ASX listed investment companies (LICs) in terms of pre-tax NTA. For further information about the Enlarged Group see Section 6.

The estimated combined pre-tax NTA of the Enlarged Group does not include the impacts of the following:

- 4,752,220 new CIE Shares issued on 16 September 2020 under the SPP to existing CIE Shareholders at \$0.625 per CIE Share, raising \$3.0m; and
- acquisition of Concentrated Leaders Fund Limited by WAM through an off-market takeover offer scheduled to close on 19 October 2020 unless extended or withdrawn. As at 31 August 2020, CLF's pre-tax NTA was approximately \$65.5m.

On 3 September 2020, WAM lodged a bidder's statement and announced its intention to acquire all of the issued shares of Concentrated Leaders Fund Limited (ASX: CLF) by way of a conditional off-market takeover offer (**CLF Offer**). The CLF Offer is currently open and will close on 19 October 2020 (unless extended or withdrawn). If the CLF Offer is successful, CLF will also become a wholly owned subsidiary of WAM and will form part of the Enlarged Group. See www.asx.com.au for details of the CLF Offer. The CLF Offer is not considered to be a material transaction for the purposes of this Bidder's Statement.

Under the CLF Offer, the maximum number of WAM Shares which would be issued as consideration is 28,290,524.

WAM's lower management expense ratio

Apart from general corporate expenses, the main expenses that each of WAM and CIE incur as LICs are investment management expenses. These expenses, paid to WAM's and CIE's respective investment managers (excluding GST), are set out below:

- WAM management fee 1.0%; and
- CIE management fee 1.4%.

The CIE management fee is 40% higher than that of WAM. The historical management expense ratios⁴ for 2019 and 2020 of each WAM and CIE, are set out below:

• 2019: WAM 1.1%% vs CIE 1.8%

• 2020: WAM 1.1% vs CIE 1.9%

As a result of the new CIE Investment Management Agreement, CIE Shareholders will be subject to a higher management fee and it is expected that CIE Shareholders will incur a management expense ratio of approximately 2.2% going forward, compared to WAM's management expense ratio of 1.1%.

b) Utilise WAM's greater on-market liquidity to exit their position

WAM has greater on-market liquidity than CIE. The Enlarged Group will also have a substantially greater market capitalisation than CIE. This may provide further increased levels of liquidity in contrast to the liquidity of CIE Shares.

As a result, CIE Shareholders who do not wish to become long-term WAM Shareholders can accept the Offer and utilise WAM's liquidity to exit their position at a premium to both the CIE Share price and the CIE pre-tax NTA.⁵

-

⁴ The management expense ratios are calculated by comparing company related expenses, together with the management fee, to the average net assets over the period. The calculation excludes portfolio related expenses such as finance costs, dividends on borrowed stock and brokerage.

⁵ CIE Shareholders are being offered consideration under the Offer that comprises a specified number of WAM Shares, rather than a number of WAM Shares with a specified market value. As a result, the value of the consideration, and any premium as a result, will fluctuate depending on the value of WAM Shares, the value of CIE Shares and the value of CIE's underlying pre-tax NTA.

Key Questions

This Section answers some key questions that you may have about the Offer and should only be read in conjunction with the entire Bidder's Statement.

1.1. What is the Bidder's Statement?

This Bidder's Statement sets out the terms of WAM's Offer and information relating to the Offer and the consideration you will receive.

1.2. What is the Offer?

WAM is offering to acquire all of the CIE Shares in which WAM does not already have a Relevant Interest, by way of an off-market takeover offer.

The Offer extends to CIE Shares on issue at 7.00pm (Sydney time) on 30 September 2020 (**Register Date**) and any CIE Shares issued during the Offer Period.

The Offer consideration is 1 WAM Share for every 3 CIE Shares you hold. The terms of the Offer are set out in Schedule 1 of this Bidder's Statement.

1.3. What is the Value of the Offer?

Based on the closing price of WAM Shares on the ASX on 25 September 2020 (the last trading day before the date of this Bidder's Statement) of \$2.24, the implied Offer value is approximately \$0.747 per CIE Share, a premium of 17.6% to the closing price of CIE Shares.

1.4. How do I accept the Offer?

CHESS Holdings

If your CIE Shares are in a CHESS Holding, to accept the Offer you must either:

- a) instruct your Controlling Participant (usually your Broker) to accept the Offer on your behalf; or
- b) complete, sign and return the enclosed Acceptance Form in accordance with the instructions contained within it.

Issuer Sponsored Holdings

If your CIE Shares are held in an Issuer Sponsored Holding, to accept the Offer you must complete, sign and return the enclosed Acceptance Form in accordance with the instructions contained within it.

Participants

If you are a Participant (as defined in the ASX Settlement Rules – typically a Broker who is a participating organisation of ASX Settlement), the above does not apply. To accept the Offer you must initiate acceptance in accordance with the ASX Settlement Operating Rules.

1.5. Can I accept the Offer for part of my holding?

No, you can only accept for your entire holding. Your acceptance will be treated as being for all your CIE Shares plus any additional CIE Shares registered as held by you at the date your acceptance is processed.

1.6. If I accept the Offer will I receive the WAM Dividend?

Yes, if you accept the Offer and are issued WAM Shares under the Offer before the WAM Dividend record date on 20 November 2020, and you continue to hold your WAM Shares on that date, you will also receive the WAM Dividend. For further information about when you will be issued WAM Shares see section 1.10.

If you accept the Offer and are issued WAM Shares under the Offer before the WAM Dividend record date on 20 November 2020, and you continue to hold your WAM Shares on that date, you will also receive the fully franked WAM Dividend of 7.75 cents per WAM Share announced on 8 July 2020 and payable on 27 November 2020 (WAM Dividend).

If the Offer is extended, and does not become unconditional, you may not receive your WAM Shares in time to receive the WAM Dividend. If the Offer does not become unconditional and is extended for less than one month, you will not be able to withdraw your acceptance of the Offer, and you will not receive the WAM Dividend. In order to receive the WAM Dividend, it is important that you accept the Offer as early as possible and in any event on or before 13 November 2020 to receive your WAM Shares by 20 November 2020.

CIE Shareholders are being offered consideration under the Offer that comprises a specified number of WAM Shares, rather than a number of WAM Shares with a specified market value. As a result, the value of the consideration will fluctuate depending on the value of WAM Shares.

For further information about when you will be issued WAM shares see section 1.10.

1.7. Are there any conditions to the Offer?

Yes, the Offer is subject to the conditions detailed at clause 8.1 of Schedule 1 of this Bidder's Statement, and summarised below – refer to Schedule 1 clause 8 for more detail:

- a) before the end of the Offer Period, WAM has a Relevant Interest in at least 50.1% (by number) of CIE Shares;
- b) between the date of this Bidder's Statement and the end of the Offer Period the Investment Management Agreement is not varied or assigned and no new investment management agreement is entered into (or CIE enters into an agreement in relation to any of the foregoing) other than assignment to, or entry into an agreement with, Wilson Asset Management;
- c) between the date of this Bidder's Statement and the end of the Offer Period no Prescribed Occurrence occurs;
- d) the pre-tax NTA of CIE does not decline by 7.5% or more below the pre-tax net NTA of CIE of \$0.746 per share, announced to the ASX in the Monthly NTA Statement 31 August 2020 on 4 September 2020;
- e) the Dow Jones Industrial Average, in AUD terms, does not close at a level that is 5% or more below the level of that index at 5.00pm (Sydney time) on the trading day immediately prior to the Announcement Date and remaining at or below that level for at least two consecutive trading days;

- between the date of this Bidder's Statement and the end of the Offer Period there is no material adverse change to the assets, liabilities, financial position, performance, profitability or prospects of CIE and/or its Subsidiaries;
- g) between the date of this Bidder's Statement and the end of the Offer Period CIE and/or its Subsidiaries do not undertake certain actions such as making or proposing to make any variations to its constitution, giving or agreeing to give any encumbrance over any of its assets, or appointing any additional Director to its Board of Directors; and
- h) between the date of this Bidder's Statement and the end of the Offer Period there is no regulatory action which materially affects the Offer.

Please see Schedule 1 for further details of the Conditions to the Offer. If the Conditions are not satisfied before the Offer closes or waived by WAM the Offer will lapse. WAM will make an announcement to the ASX if the Conditions are satisfied or waived during the Offer Period.

1.8. What happens if I accept the Offer?

Once you accept the Offer in respect of all of your CIE Shares, you will not be able to sell those CIE Shares or otherwise deal with the Rights attaching to those CIE Shares subject to your limited statutory rights to withdraw your acceptance in certain circumstances.

If you accept the Offer and are issued WAM Shares under the Offer before the WAM Dividend record date on 20 November 2020, and you continue to hold your WAM Shares on that date, you will also receive the fully franked WAM Dividend of 7.75 cents per WAM Share announced on 8 July 2020 and payable on 27 November 2020 (WAM Dividend).

If the Offer is extended, and does not become unconditional, you may not receive your WAM Shares in time to receive the WAM Dividend. If the Offer does not become unconditional and is extended for less than one month, you will not be able to withdraw your acceptance of the Offer, and you will not receive the WAM Dividend. In order to receive the WAM Dividend, it is important that you accept the Offer as early as possible and in any event on or before 13 November 2020 to receive your WAM Shares by 20 November 2020.

CIE Shareholders are being offered consideration under the Offer that comprises a specified number of WAM Shares, rather than a number of WAM Shares with a specified market value. As a result, the value of the consideration will fluctuate depending on the value of WAM Shares.

1.9. What happens if I do not accept the Offer?

Subject to what is stated below, you will remain the holder of your CIE Shares if you do not accept the Offer.

If WAM obtains a Relevant Interest in at least 90% of the CIE Shares (by number) on issue at any time during the Offer Period and the Defeating Conditions are satisfied or waived, WAM presently intends to proceed to compulsorily acquire your CIE Shares if you have not accepted the Offer (see Section 7.3). At the conclusion of the compulsory acquisition process, you will receive 1 WAM Share for every 3 CIE Shares which are compulsorily acquired. You will receive consideration for your CIE Shares sooner if you accept the Offer.

If WAM obtains a Relevant Interest in less than 90% of CIE Shares, WAM will not be able to proceed to compulsorily acquire your CIE Shares. WAM's intentions if it has a Relevant Interest in less than 90% of CIE Shares are described in Section 7.

If you do not accept the Offer, you will not receive the fully franked WAM Dividend of 7.75 cents per WAM Share announced on 8 July 2020 and payable on 27 November 2020 (WAM Dividend).

1.10. If I accept the Offer, when will I receive my WAM Shares?

If you accept this Offer, WAM will, in the usual course, issue the WAM Shares to which you are entitled on or before the earlier of:

- the day one month after you accept this Offer or, if this Offer is subject to Conditions when accepted, one month after the contract resulting from your acceptance becomes unconditional; and
- the day 21 days after the end of the Offer Period.

Full details of when you will be issued the consideration are set out in Schedule 1. It is WAM's intention however to issue the Offer Consideration within seven days of the end of the Offer Period (provided that the Offer is then unconditional).

1.11. Am I entitled to receive any CIE dividend declared?

If a dividend is declared or paid by CIE after the Announcement Date, and you receive the dividend, or are issued CIE shares under the CIE DRP, the consideration issued to you will be reduced. See clauses 7.6 and 7.7 of Schedule 1.

1.12. Can I withdraw my acceptance?

Under the terms of the Offer, you cannot withdraw your acceptance unless a withdrawal right arises under the Corporations Act. Such a withdrawal right will arise if, after you accept the Offer, WAM varies the Offer in a way that postpones for more than one month from the time when WAM has to meet its obligations under the Offer.

In the event that the Offer is extended for any reason, but for less than one month, you may not receive your WAM Shares in time to receive the WAM Dividend and you will not be able to withdraw from the Offer.

1.13. When does the Offer close?

The Offer is currently scheduled to close at 7:00 pm (Sydney time) on 13 November 2020, unless extended or withdrawn.

1.14. Can WAM extend the Offer Period?

Yes, the Offer can be extended by WAM or otherwise in accordance with the Corporations Act. You will be sent written notice of any extension, as required by the Corporations Act.

1.15. What rights will my WAM Shares have?

The WAM Shares issued under the Offer will be fully paid and will rank equally for dividends with existing WAM Shares and have the same rights (including voting rights) as existing WAM Shares.

A detailed explanation of the rights and liabilities attaching to WAM shares is set out in Section 11.3.

1.16. Will my WAM Shares be listed on the ASX?

An application will be made within seven days after the date of this Bidder's Statement to the ASX for Official Quotation of the WAM Shares issued under the Offer. Quotation of the WAM Shares to be issued under the Offer will not be automatic but will depend on the ASX exercising its discretion to admit them to the Official List.

WAM is already admitted to the Official List of ASX and shares in WAM in the same class or on the same terms as those to be issued under the Offer are already quoted.

1.17. What happens if I accept the Offer?

If you accept the Offer and it becomes or is declared unconditional, you will be issued WAM Shares within the specified timeframe.

Once you accept the Offer (even while it remains subject to Defeating Conditions) you will not be able to sell your CIE Shares on market or otherwise deal with the Rights attaching to your CIE Shares, subject to your limited statutory rights to withdraw your acceptance in certain circumstances.

1.18. What happens if I do not accept the Offer?

If you do not accept the Offer, you will remain a CIE Shareholder and will not be issued WAM Shares.

However, if WAM becomes entitled to compulsorily acquire your CIE Shares, it intends to do so. If this occurs, you will still receive the Offer Consideration for your CIE Shares but at a later date than you would have received it if you had accepted the Offer. If you receive your WAM Shares as Offer Consideration under a compulsory acquisition, this will occur after 20 November 2020 and you will not receive the WAM Dividend.

1.19. What are the tax implications of accepting the Offer?

A general description of the taxation treatment for certain Australian resident CIE Shareholders accepting the Offer is set out in Section 10.

You should consult your taxation adviser for detailed taxation advice before deciding whether or not to accept the Offer for your CIE Shares.

CIE Shareholders should be aware that if WAM does not acquire ownership of at least 80% of CIE Shares, then rollover relief will not be available to CIE Shareholders in respect of any capital gain that is made on the receipt of WAM Shares as consideration.

1.20. What are the significant risks of the Offer?

You should carefully consider the risk factors that could affect the performance of WAM and the Enlarged Group before deciding whether to accept the Offer. Many of these risks are outside the control of WAM and its Board of Directors, and cannot be mitigated. A summary of the key risks is set out in Section 8.

1.21. Will I need to pay brokerage?

On acceptance of this Offer, if you have an Issuer Sponsored Holding you will exchange your CIE Shares for WAM Shares without incurring brokerage fees.

If your CIE Shares are held in a CHESS Holding or you hold your CIE Shares through a bank, custodian or other nominee, you should ask your Controlling Participant (usually your Broker or the bank, custodian or other nominee) whether it will charge you any transaction fees or service charges connected with you accepting the Offer.

If you are a Foreign Shareholder, the cash proceeds that you will be sent (following the sale of WAM Shares that you would otherwise have been entitled to receive under the Offer) will be net of transaction costs and will be paid by cheque in Australian dollars.

If you choose not to accept the Offer, should you wish to exchange your CIE Shares for WAM Shares, you may be required to sell your shares on market, and use the proceeds to purchase WAM Shares. You may ordinarily be required to pay brokerage fees on both the sale and the purchase of shares.

1.22. Will I need to pay stamp duty?

There should be no stamp duty payable by CIE Shareholders upon the issue to them of WAM Shares. Should any stamp duty be payable on the transfer of CIE Shares to WAM pursuant to the Offer due to a change in the relevant legislation during the Offer Period, these costs will be paid by WAM.

1.23. What happens if CIE holds the requisitioned WAM EGM?

On 11 September 2020, the WAM Entities lodged a notice under section 249D of the Corporations Act, requesting that CIE call and arrange the WAM EGM at which WAM would seek to put a resolution to members to remove Mark Kerr and Don Clarke as CIE Directors, and appoint Geoff Wilson AO and Glenn Burge in their place, with a view to subsequently appointing Wilson Asset Management as CIE's new investment manager. Wilson Asset Management is also the manager of WAM.

CIE has not yet called the WAM EGM and has until 2 October 2020 to do so. Depending on the date CIE sets for the WAM EGM, the Offer may have already closed by the time the WAM EGM is held. See Section 7 for details of WAM's intentions.

If the WAM EGM occurs during the Offer Period, CIE Shareholders will have the opportunity to vote on resolutions to appoint Geoff Wilson AO and Glenn Burge to the CIE Board before the Offer closes.

2. About WAM Capital Limited

2.1. Corporate

WAM Capital Limited (ASX: WAM), one of Australia's leading LICs, is managed by Wilson Asset Management. Listed in August 1999, WAM provides investors with exposure to an actively managed, diversified portfolio of undervalued growth companies listed on the ASX, with a focus on small-to-medium sized businesses. WAM's investment objectives are to deliver a stream of fully franked dividends, provide capital growth and preserve capital.

2.2. WAM Directors

The WAM Board is:

- a) Geoff Wilson AO (Chairman);
- b) Kate Thorley;
- c) Dr Philippa Ryan;
- d) James Chirnside;
- e) Lindsay Mann; and
- f) Matthew Pancino.

2.3. Overview of WAM's historical financial performance

A summary of WAM's historical financial performance, financial position and cash flows for the financial years ended 30 June 2020, 30 June 2019 and 30 June 2018 is set out below:

Statement of Financial Performance	30 June 2020	30 June 2019	30 June 2018
	\$	\$	\$
Net realised and unrealised gains/(losses) on financial assets	(56,631,902)	(18,324,114)	144,556,373
Other revenue from operating activities	28,967,203	44,114,382	43,640,946
Management fees	(13,118,338)	(13,682,297)	(13,507,291)
Performance fees	-	-	(339,457)
Directors fees	(140,710)	(150,000)	(150,109)
Brokerage expense on share purchases	(4,811,676)	(4,251,736)	(5,174,373)
Expenses paid on borrowed stock	(138,909)	(93,367)	(930,945)
Custody fees	(107,464)	(101,841)	(98,734)
ASX listing and chess fees	(228,690)	(224,441)	(213,988)
Share registry fees	(291,044)	(315,925)	(216,965)
Disbursements, mailing and printing	(280,782)	(272,441)	(211,018)
Legal and professional fees	(41,307)	(170,203)	(112,069)
ASIC industry funding levy	(54,042)	(112,582)	-
Other expenses from ordinary activities	(371,378)	(427,834)	(375,581)
Profit/(loss) before income tax	(47,249,039)	5,987,601	166,866,789
Income tax expense/(benefit)	20,542,072	8,545,621	(41,469,154)
Profit/(loss) after income tax attributable to members of the Company	(26,706,967)	14,533,222	125,397,635
Other comprehensive income/(loss)			
Other comprehensive income/(loss) for the year, net of tax	-	-	-
Total comprehensive income/(loss) for the year	(26,706,967)	14,533,222	125,397,635
Basic and diluted earnings/(loss) per share	(3.70 cents)	2.08 cents	19.30 cents

30 June 2020 30 June 2019 30 June 2018

Statement of Financial Position	30 Julie 2020	30 Julie 2013	30 Julie 2010
	\$	\$	\$
Current assets			
Cash and cash equivalents	203,400,924	325,988,043	389,675,066
Trade and other receivables	20,984,982	37,858,074	33,758,932
Financial assets	963,612,288	1,000,725,212	973,929,383
Current tax assets	22,676,280	-	-
Total current assets	1,210,674,474	1,364,571,329	1,397,363,381
Non-current assets			
Deferred tax assets	68,540,392	23,613,806	2,598,344
Total non-current assets	68,540,392	23,613,806	2,598,344
Total assets	1,279,214,866	1,388,185,135	1,399,961,725
Current liabilities			
Trade and other payables	32,857,926	31,424,411	31,383,915
Financial liabilities	897,913	10,049,817	9,851,683
Current tax liabilities	-	2,293,796	20,094,479
Total current liabilities	33,755,839	43,768,024	61,330,077
Non-current liabilities			
Deferred tax liabilities	23,778,042	-	14,798,434
Total non-current liabilities	23,778,042	-	14,798,434
Total liabilities	57,533,881	43,768,024	76,128,511
Net assets	1,221,680,985	1,344,417,111	1,323,833,214
Equity			
Issued capital	1,352,964,191	1,337,451,236	1,221,306,473
Profits reserve	44,600,021	54,582,036	130,146,739
Accumulated losses	(175,883,227)	(47,616,161)	(27,619,998)

Total equity

1,221,680,985 1,344,417,111 1,323,833,214

\$	Ś	\$
S	S	- 5
Y	Y	Y

Cash flows from operating activities Proceeds from sale of investments 3,724,169,098 3,100,686,678 3,206,321,323 Payments for purchase of investments (3,736,901,301) (3,052,055,397) (3,209,713,092) Dividends received 23,775,239 32,001,949 27,955,581 Interest received 3,175,482 7,329,164 7,468,206 Other investment income received 3,540,505 5,330,021 7,391,017 Management fee (GST inclusive) (14,233,258) (14,713,922) (14,386,876) Performance fee (GST inclusive) (5,155,458) (4,556,885) (5,552,654) Payments for administration expenses (GST inclusive) (1,821,994) (1,853,707) (2,538,859) Income tax paid (24,471,376) (4,2959,890) (32,021,890) GST on brokerage expense on share sales (364,037) (309,398) (369,914) Net CST received from ATO 1,729,140 1,734,661 1,794,165 Net Cash provided by/(used in) operating activities (26,557,960) 30,288,979 (13,652,993) Cash flows from financing activities (96,029,159) (93,8		Y	~	Y
Payments for purchase of investments (3,736,901,301) (3,052,055,397) (3,209,713,092) Dividends received 23,775,239 32,001,949 27,955,581 Interest received 3,175,482 7,329,164 7,468,206 Other investment income received 3,540,505 5,330,021 7,391,017 Management fee (GST inclusive) (14,233,258) (14,713,922) (14,386,876) Performance fee (GST inclusive) - (364,295) - Brokerage expense on share purchases (GST inclusive) (5,155,458) (4,556,885) (5,552,654) Payments for administration expenses (GST inclusive) (1,821,994) (1,853,707) (2,538,859) Income tax paid (24,471,376) (42,959,890) (32,021,890) GST on brokerage expense on share sales (364,037) (309,398) (369,914) Net GST received from ATO 1,729,140 1,734,661 1,794,165 Net cash provided by/(used in) operating activities (26,557,960) 30,268,979 (13,652,993) Cash flows from financing activities (96,029,159) (93,856,468) (82,072,950) Share issue costs (99	Cash flows from operating activities			
Dividends received 23,775,239 32,001,949 27,955,581 Interest received 3,175,482 7,329,164 7,468,206 Other investment income received 3,540,505 5,330,021 7,391,017 Management fee (GST inclusive) (14,233,258) (14,713,922) (14,386,876) Performance fee (GST inclusive) - (364,295) -	Proceeds from sale of investments	3,724,169,098	3,100,686,678	3,206,321,323
Interest received 3,175,482 7,329,164 7,468,206 Other investment income received 3,540,505 5,330,021 7,391,017 Management fee (GST inclusive) (14,233,258) (14,713,922) (14,386,876) Performance fee (GST inclusive) - (364,295) - (364,037) - (309,398) - (369,914) - (369,039) - (369,914) - (369,039) - (369,914) - (369,039) - (369,	Payments for purchase of investments	(3,736,901,301)	(3,052,055,397)	(3,209,713,092)
Other investment income received 3,540,505 5,330,021 7,391,017 Management fee (GST inclusive) (14,233,258) (14,713,922) (14,386,876) Performance fee (GST inclusive) - (364,295) - Brokerage expense on share purchases (GST inclusive) (5,155,458) (4,556,885) (5,552,654) Payments for administration expenses (GST inclusive) (1,821,994) (1,853,707) (2,538,859) Income tax paid (24,471,376) (42,959,890) (32,021,890) GST on brokerage expense on share sales (364,037) (309,398) (369,914) Net GST received from ATO 1,729,140 1,734,661 1,794,165 Net cash provided by/(used in) operating activities (26,557,960) 30,268,979 (13,652,993) Cash flows from financing activities 9(9,029,159) (93,856,468) (82,072,950) Share issue costs - (99,534) (779,542) Net cash provided by/(used in) financing activities (96,029,159) (93,956,002) (16,696,742) Net increase/(decrease) in cash and cash equivalents held (122,587,119) (63,687,023) (30,349,735)	Dividends received	23,775,239	32,001,949	27,955,581
Management fee (GST inclusive) (14,233,258) (14,713,922) (14,386,876) Performance fee (GST inclusive) - (364,295) - Brokerage expense on share purchases (GST inclusive) (5,155,458) (4,556,885) (5,552,654) Payments for administration expenses (GST inclusive) (1,821,994) (1,853,707) (2,538,859) Income tax paid (24,471,376) (42,959,890) (32,021,890) GST on brokerage expense on share sales (364,037) (309,398) (369,914) Net GST received from ATO 1,729,140 1,734,661 1,794,165 Net cash provided by/(used in) operating activities (26,557,960) 30,268,979 (13,652,993) Cash flows from financing activities 96,029,159) (93,856,468) (82,072,950) Dividends paid – net of reinvestment (96,029,159) (93,856,468) (82,072,950) Share issue costs - (99,534) (779,542) Net cash provided by/(used in) financing activities (96,029,159) (93,956,002) (16,696,742) Net increase/(decrease) in cash and cash equivalents held (122,587,119) (63,687,023) (30,349,735) Cash and cash equivalents at beginning of financial year 203	Interest received	3,175,482	7,329,164	7,468,206
Performance fee (GST inclusive) - (364,295) - Brokerage expense on share purchases (GST inclusive) (5,155,458) (4,556,885) (5,552,654) Payments for administration expenses (GST inclusive) (1,821,994) (1,853,707) (2,538,859) Income tax paid (24,471,376) (42,959,890) (32,021,890) GST on brokerage expense on share sales (364,037) (309,398) (369,914) Net GST received from ATO 1,729,140 1,734,661 1,794,165 Net cash provided by/(used in) operating activities (26,557,960) 30,268,979 (13,652,993) Cash flows from financing activities Proceeds from issue of shares - - 66,155,750 Dividends paid - net of reinvestment (96,029,159) (93,856,468) (82,072,950) Share issue costs - (99,534) (779,542) Net cash provided by/(used in) financing activities (96,029,159) (93,956,002) (16,696,742) Net increase/(decrease) in cash and cash equivalents held (122,587,119) (63,687,023) (30,349,735) Cash and cash equivalents at end of financia	Other investment income received	3,540,505	5,330,021	7,391,017
Brokerage expense on share purchases (GST inclusive) (5,155,458) (4,556,885) (5,552,654) Payments for administration expenses (GST inclusive) (1,821,994) (1,853,707) (2,538,859) Income tax paid (24,471,376) (42,959,890) (32,021,890) GST on brokerage expense on share sales (364,037) (309,398) (369,914) Net GST received from ATO 1,729,140 1,734,661 1,794,165 Net cash provided by/(used in) operating activities (26,557,960) 30,268,979 (13,652,993) Cash flows from financing activities 9,029,159 (93,856,468) (82,072,950) Dividends paid – net of reinvestment (96,029,159) (93,856,468) (82,072,950) Share issue costs - (99,534) (779,542) Net cash provided by/(used in) financing activities (96,029,159) (93,956,002) (16,696,742) Net increase/(decrease) in cash and cash equivalents held (122,587,119) (63,687,023) (30,349,735) Cash and cash equivalents at beginning of financial year 325,988,043 389,675,066 420,024,801 Non-cash transactions: Shares issued via dividend reinvestment plan 15,512,955 16,237,620 16,210,218 <td>Management fee (GST inclusive)</td> <td>(14,233,258)</td> <td>(14,713,922)</td> <td>(14,386,876)</td>	Management fee (GST inclusive)	(14,233,258)	(14,713,922)	(14,386,876)
Payments for administration expenses (GST inclusive) (1,821,994) (1,853,707) (2,538,859) Income tax paid (24,471,376) (42,959,890) (32,021,890) GST on brokerage expense on share sales (364,037) (309,398) (369,914) Net GST received from ATO 1,729,140 1,734,661 1,794,165 Net cash provided by/(used in) operating activities (26,557,960) 30,268,979 (13,652,993) Cash flows from financing activities - - 66,155,750 Dividends paid – net of reinvestment (96,029,159) (93,856,468) (82,072,950) Share issue costs - (99,534) (779,542) Net cash provided by/(used in) financing activities (96,029,159) (93,956,002) (16,696,742) Net increase/(decrease) in cash and cash equivalents held (122,587,119) (63,687,023) (30,349,735) Cash and cash equivalents at beginning of financial year 325,988,043 389,675,066 420,024,801 Cash and cash equivalents at end of financial year 203,400,924 325,988,043 389,675,066 Non-cash transactions: 5 16,210,218	Performance fee (GST inclusive)	-	(364,295)	-
Income tax paid	Brokerage expense on share purchases (GST inclusive)	(5,155,458)	(4,556,885)	(5,552,654)
GST on brokerage expense on share sales (364,037) (309,398) (369,914) Net GST received from ATO 1,729,140 1,734,661 1,794,165 Net cash provided by/(used in) operating activities (26,557,960) 30,268,979 (13,652,993) Cash flows from financing activities Proceeds from issue of shares - - 66,155,750 Dividends paid - net of reinvestment (96,029,159) (93,856,468) (82,072,950) Share issue costs - (99,534) (779,542) Net cash provided by/(used in) financing activities (96,029,159) (93,956,002) (16,696,742) Net increase/(decrease) in cash and cash equivalents held (122,587,119) (63,687,023) (30,349,735) Cash and cash equivalents at beginning of financial year 325,988,043 389,675,066 420,024,801 Cash and cash equivalents at end of financial year 203,400,924 325,988,043 389,675,066 Non-cash transactions: Shares issued via dividend reinvestment plan 15,512,955 16,237,620 16,210,218	Payments for administration expenses (GST inclusive)	(1,821,994)	(1,853,707)	(2,538,859)
Net GST received from ATO 1,729,140 1,734,661 1,794,165 Net cash provided by/(used in) operating activities (26,557,960) 30,268,979 (13,652,993) Cash flows from financing activities Proceeds from issue of shares - - 66,155,750 Dividends paid - net of reinvestment (96,029,159) (93,856,468) (82,072,950) Share issue costs - (99,534) (779,542) Net cash provided by/(used in) financing activities (96,029,159) (93,956,002) (16,696,742) Net increase/(decrease) in cash and cash equivalents held (122,587,119) (63,687,023) (30,349,735) Cash and cash equivalents at beginning of financial year 325,988,043 389,675,066 420,024,801 Non-cash transactions: Shares issued via dividend reinvestment plan 15,512,955 16,237,620 16,210,218	Income tax paid	(24,471,376)	(42,959,890)	(32,021,890)
Net cash provided by/(used in) operating activities (26,557,960) 30,268,979 (13,652,993) Cash flows from financing activities Proceeds from issue of shares - 66,155,750 Dividends paid – net of reinvestment (96,029,159) (93,856,468) (82,072,950) Share issue costs - (99,534) (779,542) Net cash provided by/(used in) financing activities (96,029,159) (93,956,002) (16,696,742) Net increase/(decrease) in cash and cash equivalents held (122,587,119) (63,687,023) (30,349,735) Cash and cash equivalents at beginning of financial year 325,988,043 389,675,066 420,024,801 Cash and cash equivalents at end of financial year 203,400,924 325,988,043 389,675,066 Non-cash transactions: Shares issued via dividend reinvestment plan 15,512,955 16,237,620 16,210,218	GST on brokerage expense on share sales	(364,037)	(309,398)	(369,914)
Cash flows from financing activities Proceeds from issue of shares - - 66,155,750 Dividends paid – net of reinvestment (96,029,159) (93,856,468) (82,072,950) Share issue costs - (99,534) (779,542) Net cash provided by/(used in) financing activities (96,029,159) (93,956,002) (16,696,742) Net increase/(decrease) in cash and cash equivalents held (122,587,119) (63,687,023) (30,349,735) Cash and cash equivalents at beginning of financial year 325,988,043 389,675,066 420,024,801 Cash and cash equivalents at end of financial year 203,400,924 325,988,043 389,675,066 Non-cash transactions: Shares issued via dividend reinvestment plan 15,512,955 16,237,620 16,210,218	Net GST received from ATO	1,729,140	1,734,661	1,794,165
Proceeds from issue of shares - - 66,155,750 Dividends paid – net of reinvestment (96,029,159) (93,856,468) (82,072,950) Share issue costs - (99,534) (779,542) Net cash provided by/(used in) financing activities (96,029,159) (93,956,002) (16,696,742) Net increase/(decrease) in cash and cash equivalents held (122,587,119) (63,687,023) (30,349,735) Cash and cash equivalents at beginning of financial year 325,988,043 389,675,066 420,024,801 Cash and cash equivalents at end of financial year 203,400,924 325,988,043 389,675,066 Non-cash transactions: Shares issued via dividend reinvestment plan 15,512,955 16,237,620 16,210,218	Net cash provided by/(used in) operating activities	(26,557,960)	30,268,979	(13,652,993)
Dividends paid – net of reinvestment (96,029,159) (93,856,468) (82,072,950) Share issue costs - (99,534) (779,542) Net cash provided by/(used in) financing activities (96,029,159) (93,956,002) (16,696,742) Net increase/(decrease) in cash and cash equivalents held (122,587,119) (63,687,023) (30,349,735) Cash and cash equivalents at beginning of financial year 325,988,043 389,675,066 420,024,801 Cash and cash equivalents at end of financial year 203,400,924 325,988,043 389,675,066 Non-cash transactions: Shares issued via dividend reinvestment plan 15,512,955 16,237,620 16,210,218	Cash flows from financing activities			
Dividends paid – net of reinvestment (96,029,159) (93,856,468) (82,072,950) Share issue costs - (99,534) (779,542) Net cash provided by/(used in) financing activities (96,029,159) (93,956,002) (16,696,742) Net increase/(decrease) in cash and cash equivalents held (122,587,119) (63,687,023) (30,349,735) Cash and cash equivalents at beginning of financial year 325,988,043 389,675,066 420,024,801 Cash and cash equivalents at end of financial year 203,400,924 325,988,043 389,675,066 Non-cash transactions: Shares issued via dividend reinvestment plan 15,512,955 16,237,620 16,210,218				66 155 750
Share issue costs - (99,534) (779,542) Net cash provided by/(used in) financing activities (96,029,159) (93,956,002) (16,696,742) Net increase/(decrease) in cash and cash equivalents held (122,587,119) (63,687,023) (30,349,735) Cash and cash equivalents at beginning of financial year 325,988,043 389,675,066 420,024,801 Cash and cash equivalents at end of financial year 203,400,924 325,988,043 389,675,066 Non-cash transactions: Shares issued via dividend reinvestment plan 15,512,955 16,237,620 16,210,218	Dividends paid - net of reinvestment	(96,029,159)	(93,856,468)	
Net cash provided by/(used in) financing activities (96,029,159) (93,956,002) (16,696,742) Net increase/(decrease) in cash and cash equivalents held (122,587,119) (63,687,023) (30,349,735) Cash and cash equivalents at beginning of financial year 325,988,043 389,675,066 420,024,801 Cash and cash equivalents at end of financial year 203,400,924 325,988,043 389,675,066 Non-cash transactions: Shares issued via dividend reinvestment plan 15,512,955 16,237,620 16,210,218		-		
Cash and cash equivalents at beginning of financial year 325,988,043 389,675,066 420,024,801 Cash and cash equivalents at end of financial year 203,400,924 325,988,043 389,675,066 Non-cash transactions: Shares issued via dividend reinvestment plan 15,512,955 16,237,620 16,210,218	Net cash provided by/(used in) financing activities	(96,029,159)		
Cash and cash equivalents at end of financial year 203,400,924 325,988,043 389,675,066 Non-cash transactions: Shares issued via dividend reinvestment plan 15,512,955 16,237,620 16,210,218	Net increase/(decrease) in cash and cash equivalents held	(122,587,119)	(63,687,023)	(30,349,735)
Non-cash transactions: Shares issued via dividend reinvestment plan 15,512,955 16,237,620 16,210,218	Cash and cash equivalents at beginning of financial year	325,988,043	389,675,066	420,024,801
Shares issued via dividend reinvestment plan 15,512,955 16,237,620 16,210,218	Cash and cash equivalents at end of financial year	203,400,924	325,988,043	389,675,066
·	Non-cash transactions:			
Shares issued via scrip consideration for acquisition - 99,976,817 -	Shares issued via dividend reinvestment plan	15,512,955	16,237,620	16,210,218
	Shares issued via scrip consideration for acquisition	-	99,976,817	-

Source: The above information has been drawn from the audited financial statements for WAM and associated disclosures included in WAM's Annual Financial Reports for the years ended 30 June 2020, 30 June 2019 and 30 June 2018.

2.4. Public announcements by WAM

WAM is a disclosing entity for the purposes of section 111AC(1) of the Corporations Act and as such, is subject to regular reporting and disclosure obligations. Broadly, these obligations require WAM to:

- a) prepare and lodge with the ASIC both annual and half-yearly financial statements accompanied by a Directors' statement and report and an audit or review report; and
- b) immediately notify the ASX of any information concerning WAM of which it is, or becomes, aware and which a reasonable person would expect to have a material effect on the price or value of securities in WAM, subject to certain limited exceptions related mainly to confidential information.

Copies of documents lodged at ASIC in relation to WAM may be obtained from or inspected at an office of ASIC. Copies of documents lodged with the ASX in relation to WAM may be obtained from the ASX's website (www.asx.com.au). In addition, you may request a copy of any such document by contacting WAM and we will provide a copy of that document free of charge.

2.5. Investment manager

About Wilson Asset Management

Established in 1997 by Geoff Wilson AO, the investment manager of WAM, Wilson Asset Management, is an independently owned investment manager and LIC specialist based in Sydney, Australia. Wilson Asset Management has a strong track record of delivering risk-adjusted returns for shareholders and making a difference for investors and the community for more than 20 years. Established in 1997 by Geoff Wilson AO, Wilson Asset Management is responsible for investing more than \$3.5 billion in Australian and international companies on behalf of 88,000 retail investors across six LICs: WAM Capital Limited, WAM Leaders Limited, WAM Global Limited, WAM Microcap Limited, WAM Research Limited, WAM Active Limited and WAM Alternative Assets Limited.

Wilson Asset Management created and is the lead supporter of the first LICs to deliver both investment and social returns: Future Generation Australia (ASX: FGX) and Future Generation Global (ASX: FGG). Wilson Asset Management advocates and acts for retail investors, is a member of the global philanthropic Pledge 1% movement and provides all team members with \$10,000 each year to donate to charities of their choice. This philanthropic investment is made by Wilson Asset Management.

WAM Capital has an investment management agreement with Wilson Asset Management, providing shareholders with the following benefits:

Risk-adjusted returns. Wilson Asset Management is focused on delivering strong returns while taking the least risk possible to protect shareholders' capital.

Diversification. LICs managed by Wilson Asset Management hold small positions in many companies, providing shareholders with access to highly diversified portfolios.

Experience. Wilson Asset Management's investment team is comprised of 14 professionals with a total focus on protecting and growing shareholder capital and is informed by more than 190 years' combined equity market experience.

Market access. Wilson Asset Management's investment team hold more than 2,500 company meetings each year, that coupled with their extensive network in the market, provides insight, intelligence and opportunities.

Transparency. Wilson Asset Management values shareholder engagement and its LICs adhere to strict corporate governance requirements.

Shareholder advocacy. Wilson Asset Management regularly and actively engages in public debate on behalf of self-managed superannuation funds and retail investors.

A positive impact. Wilson Asset Management is passionate about making a difference to the broader community.

Objectives and investment process

The investment objectives of WAM are to:

- deliver investors a stream of fully franked dividends;
- provide capital growth; and
- preserve capital.

The above is not intended to be a forecast. It is merely the Investment objectives of WAM. WAM may not be successful in meeting these objectives.

WAM provides investors with access to Wilson Asset Management's two distinctive investment processes:

- a research-driven process focused on identifying undervalued growth companies; and
- a market-driven process that takes advantage of market mispricing opportunities.

Research-driven investing

The research-driven investment process identifies undervalued growth companies through diligent and deep research that focuses on free cash flow, return on equity and the quality of a company. Each company is carefully rated with respect to management, earnings growth potential, valuation and industry position. Under this proprietary process, the investment team will only ever invest in a security once it can identify a catalyst or event that it expects will change the market's valuation of the company.



Market-driven investing

This investment process takes advantage of short-term mispricing opportunities in the Australian equity market. Opportunities are derived from initial public offerings, placements, block trades, rights issues, corporate transactions (such as takeovers, mergers, schemes of arrangement, corporate spinoffs and restructures), arbitrage opportunities, LIC discount arbitrages, short selling and trading market themes and trends.

3. Information about WAM Shares

3.1. Issued Securities

As at the date of this Bidder's Statement, there are 725,465,456 WAM Shares on issue and no other WAM securities.

On 3 September 2020, WAM lodged a bidder's statement and announced its intention to acquire all of the issued shares of Concentrated Leaders Fund Limited (ASX: CLF) by way of a conditional off-market takeover offer. The CLF Offer is currently open and will close on 19 October 2020 (unless extended or withdrawn). If the CLF Offer is successful, CLF will also become a wholly owned subsidiary of WAM and will form part of the Enlarged Group. See www.asx.com.au for details of the CLF Offer. The CLF Offer is not considered to be a material transaction for the purposes of this Bidder's Statement.

Under the CLF Offer, the maximum number of WAM Shares which would be issued as consideration is 28,290,524.

3.2. Substantial holders

As at the date of this Bidder's Statement there are no holders in WAM who have filed substantial holder notices on the ASX.

3.3. WAM Share Price

The closing price of WAM Shares on the ASX on 25 September 2020 (being the last trading day that WAM Shares traded prior to the date of this Bidder's Statement) was \$2.24.

The VWAP of WAM Shares on the ASX for the one month before the date of this Bidder's Statement was \$2.13.

The VWAP of WAM Shares on the ASX for the five trading days before the date of this Bidder's Statement was \$2.18.

The highest recorded sale price of WAM Shares on the ASX in the four months before the date this Bidder's Statement lodged with ASIC was \$2.24 on 25 September 2020.

The lowest recorded sale price of WAM Shares on the ASX in the four months before this Bidder's Statement lodged with ASIC was \$1.77 on 29 June 2020.

3.4. Rights and liabilities attached to WAM Shares offered

The WAM Shares issued as consideration under the Offer will be fully paid and rank equally with existing WAM Shares from the date of their issue.

WAM Shares are quoted on the ASX under the code 'WAM'.

An application will be made within seven days of the date of this Bidder's Statement to the ASX for Official Quotation of the WAM Shares issued under the Offer. Quotation is not guaranteed or automatic on such an application to the ASX. Nothing in this Bidder's Statement will be taken to state or imply that the WAM Shares issued under this Offer will be quoted on the ASX, but quotation is expected in the ordinary course as WAM is already admitted to the Official List of the ASX.

The rights and restrictions attaching to the WAM Shares that will be issued as the consideration, or part of the consideration, under the Offer are primarily set out in the WAM Constitution and in the Corporations Act and are subject to statutory, common law and ASX Listing Rule requirements.

4. About Contango Income Generator Limited

4.1. Disclaimer

The following description of CIE and the financial information concerning CIE contained in this Section has been prepared by WAM using publicly available information. To the extent permitted by law, WAM does not make any representation or warranty, express or implied, as to the currency, accuracy or completeness of such information.

The primary sources of information about CIE used by WAM are as follows:

- a) annual and interim financial reports issued to CIE Shareholders;
- b) announcements issued by CIE through the ASX; and
- c) CIE's website.

4.2. Overview of CIE and its principal business activities

CIE is an income focused listed investment company which invests primarily in companies within the S&P/ASX 300 Accumulation Index excluding the top 20 listed companies. CIE's Investment Manager focuses on selecting those companies that provide a consistent and growing source of dividends.

CIE was incorporated in 2012 and listed on the ASX in 2015. CIE's Investment Manager, Contango Asset Management Limited, was a wholly owned subsidiary of a listed investment company Contango MicroCap Limited (ASX: CTN) (now called NAOS Small Cap Opportunities Company Limited (ASX: NSC)). Key personnel of the CIE's Investment Manager were George Boubouras, who was the Managing Director and Chief Investment Officer of CTN and Mark Kerr, Chairman and Director of CIE and CTN.

Ownership of Contango Asset Management Limited was sold to employees of CTN and other third party investors in 2016. In 2017, George Boubouras resigned from his position as a Director of CIE following his resignation as Managing Director and Chief Investment Officer of Contango Asset Management Limited. Martin Switzer took over as the Chief Executive Officer and Managing Director of Contango Asset Management Limited. Martin Switzer has been a Director of Contango Asset Management Limited since 25 August 2016 and a Director of WCM Investment Management, LLC (WCM Investment Management) since 9 February 2017 and a Director of CIE since 21 February 2019. WCM Investment Management, LLC (WCM) is a California-based asset management firm.

In 2019, the Board of CIE provided the CIE Manager with greater flexibility to invest in growth stocks and widened the sector in which the company could invest in to beyond the S&P/ASX 300 Accumulation Index.

On 12 August 2020, CIE announced that subject to shareholder approval, CIE would adopt a new investment strategy, moving to a global long short investment strategy managed by WCM Investment Management. It is intended that the Company will access the WCM Quality Global Growth Long Short Strategy by investing in WCM Focused Global Growth Long-Short Fund.

Under the new CIE Investment Management Agreement, CIE will:

- increase its management fee payable to the CIE Manager and WCM Investment Management to 1.4% per annum (plus GST), in aggregate, of the portfolio value of CIE's; and
- introduce a performance fee of 20% (plus GST) of the portfolios outperformance relative to the 'Total Hurdle Amount', being the initial portfolio value adjusted for the benchmark (being the MSCI All Country World Index with gross dividends reinvested reported in Australian dollars and unhedged) return, plus management fees over each financial year.

If the new CIE Investment Management Agreement is terminated by the Company after the initial term of five years, by giving the CIE Manager six month's written notice, a termination fee of up to \$500,000 is payable by CIE to the CIE Manager on a sliding scale decreasing to nil if the new CIE Investment Management Agreement is terminated greater than 5 years after the Initial Term. This termination fee is in addition to any accrued but unpaid management and performance fees payable by CIE to the CIE Manager.

On 18 September 2020, 47% of shareholders who submitted their proxies voted against the proposed change in investment strategy and investment manager. The new CIE Investment Management Agreement was signed on 21 September 2020.

4.3. Directors

As at the date of this Bidder's Statement, there are three Directors of CIE, being:

- a) Mark Kerr (Chairman);
- b) Don Clarke; and
- c) Martin Switzer.

4.4. CIE substantial holders

Based on the substantial shareholder announcements by CIE released to the ASX, CIE currently has the following substantial holders:

Rank	Name	Shares	% of Shares
1	Wilson Asset Management Group	14,787,279	14.13
2	Lanyon Asset Management Pty Limited	10,701,396	8.86

4.5. CIE's performance track record

Performance at 31 July 2020	1 yr %	3 yrs %pa	Since inception %pa (Aug-2015)
Income Return ¹	4.12%	6.08%	5.53%
NTA Return ²	-23.59%	-10.52%	-6.03%
Contango Income Generator Limited ³	-19.47%	-4.44%	-0.50%

¹Income Return is the dividends attributed to the period divided by the pre-tax NTA per share at the beginning of the period. NTA is the underlying portfolio value after all fees and expenses are deducted.

Source: CIE's Investment Update and NTA Statement 31 July 2020, as announced to the ASX.

Over the 12 months to 31 July 2020, the S&P/ASX 200 Accumulation Index returned -9.9%. The CIE investment portfolio returned -19.47% for CIE Shareholders and underperformed the S&P/ASX 200 Accumulation Index by 9.6% over this period.

CIE has not released a further investment update for shareholders outside of its stated NTA per share as at 31 August 2020.

 $^{^{\}rm 2}$ NTA Return is the movement in pre-tax NTA during the period.

³ Contango Income Generator Limited return is the Income Return plus NTA Return.

4.6. Overview of CIE's historical financial performance

A summary of CIE's historical financial performance, financial position and cash flows for the financial years ended 30 June 2020, 30 June 2019 and 30 June 2018 is set out below:

Statement of Financial Performance	30 June 2020	30 June 2019	30 June 2018
Statement of Financial Performance	\$	\$	\$
Revenue and other income			
Interest income	16,000	58,000	137,000
Dividend income	5,160,000	5,220,000	4,522,000
Fair value loss on financial assets at fair value through profit or loss	(19,231,000)	(615,000)	(80,000)
	(14,055,000)	4,663,000	4,579,000
Expenses			
Investment management fees	844,000	930,000	914,000
Business administration expenses	250,000	250,000	250,000
Transaction costs	372,000	205,000	291,000
Other expenses	204,000	229,000	154,000
Listing, custody and registry costs	157,000	179,000	267,000
Legal, accounting and professional costs	141,000	93,000	48,000
Directors' remuneration expense	73,000	97,000	136,000
	2,041,000	1,983,000	2,060,000
Profit / (loss) before income tax	(16,096,000)	2,680,000	2,519,000
Income tax benefit	1,942,000	660,000	313,000
Profit / (loss) after income tax attributable to members of the Company	(14,154,000)	3,340,000	2,832,000
Other comprehensive income / (loss)			
Other comprehensive income / (loss)	-	-	-
Total comprehensive income / (loss) for the year	(14,154,000)	3,340,000	2,832,000
Basic and diluted earnings / (loss) per share	(13.52 cents)	3.21 cents	2.85 cents

Chatamant of Financial Decition	30 June 2020	30 June 2019	30 June 2018
Statement of Financial Position	\$	\$	\$
Current assets			
Cash and cash equivalents	4,052,000	5,467,000	11,999,000
Trade and other receivables	515,000	449,000	366,000
Investments at fair value through profit or loss	73,360,000	91,907,000	88,459,000
Deferred tax assets	1,467,000	_	368,000
Total assets	79,394,000	97,823,000	101,192,000
Current liabilities			
Trade and other payables	251,000	366,000	326,000
Deferred tax liabilities	-	460,000	1,512,000
Total current liabilities	251,000	826,000	1,838,000
Net assets	79,143,000	96,997,000	99,354,000
Equity			
Issued capital	102,671,000	102,228,000	101,727,000
Reserves	(23,641,000)	(5,357,000)	(5,357,000)
Retained earnings	113,000	126,000	2,984,000
Total equity	79,143,000	96,997,000	99,354,000
Statement of Cash Flows	30 June 2020 \$	30 June 2019 \$	30 June 2018 \$
Cash flows from operating activities			
Payments for purchase of investments	(62,967,000)	(34,458,000)	(57,004,000)
Proceeds from sale of investments	61,807,000	30,188,000	41,134,000
Dividends received	5,071,000	4,848,000	4,481,000
Interest received	21,000	53,000	137,000
Income tax paid	-	(39,000)	(141,000)
Other payments to suppliers and employees	(1,647,000)	(1,427,000)	(2,030,000)
Net cash provided by/(used in) operating activities	2,285,000	(835,000)	(13,423,000)
Cash flows from financing activities			
Proceeds from issue of new shares	-	-	21,914,000
Dividends paid net of amounts reinvested	(3,700,000)	(5,697,000)	(7,696,000)
Net cash provided by/(used in) financing activities	(3,700,000)	(5,697,000)	14,218,000
Net increase/(decrease) in cash and cash equivalents held	(1,415,000)	(6,532,000)	795,000
Cash and cash equivalents at beginning of financial year	5,467,000	11,999,000	11,204,000
Cash and cash equivalents at end of financial year	4,052,000	5,467,000	11,999,000

Source: The above information has been drawn from the audited financial statements for CIE and associated disclosures included in CIE's Annual Financial Reports for the years ended 30 June 2020, 30 June 2019 and 30 June 2018.

4.7. Post balance date activities

On 12 August 2020, CIE announced its intentions to call a general meeting of members to approve the new CIE Investment Management Agreement, under which CIE adopts a new investment strategy involving the investment of the Company's portfolio into the WCM Quality Global Growth Long Short Strategy.

CIE also announced that it would raise further capital to invest in this new strategy, under a Placement to professional and sophisticated investors and a SPP for existing CIE Shareholders, resulting in:

- 15,753,534 new CIE Shares being issued on 19 August 2020 under the Placement to professional and sophisticated investors at \$0.625 per CIE Share, raising \$9.8m; and
- 4,752,220 new CIE Shares being issued on 16 September 2020 under the SPP to existing CIE Shareholders at \$0.625 per CIE Share, raising \$3.0m.

Both the Placement and the SPP were conducted at substantial discounts to CIE's pre-tax NTA, diluting existing CIE Shareholders.

4.8. ASX announcements

CIE is a disclosing entity for the purposes of section 111AC(1) of the Corporations Act and as such, is subject to regular reporting and disclosure obligations. Broadly, these obligations require CIE to:

- a) prepare and lodge with ASIC both annual and half-year financial statements accompanied by a Directors' statement and report and an audit or review report; and
- b) immediately notify the ASX of any information concerning CIE of which it is, or becomes, aware and which a reasonable person would expect to have a material effect on the price or value of securities in CIE, subject to certain limited exceptions related mainly to confidential information.

Copies of documents lodged at ASIC in relation to CIE may be obtained from or inspected at an office of ASIC. Copies of documents lodged with the ASX in relation to CIE may be obtained from the ASX's website (www.asx.com.au), or inspected at the offices of the ASX.

Since 30 June 2020, CIE has issued the following announcements to the ASX:

Date	Announcement
21/09/2020	New Investment Management Agreement
18/09/2020	Change of Director's Interest Notice
18/09/2020	Change of Director's Interest Notice
18/09/2020	Change of Director's Interest Notice
18/09/2020	Results of Meeting
18/09/2020	Chairman's Address to Shareholders
16/09/2020	Receipt of shareholder notice
16/09/2020	Appendix 2A
15/09/2020	Completion of Share Purchase Plan - 15 September 2020
11/09/2020	Change in substantial holding
10/09/2020	Shareholder Update - 10 September 2020
4/09/2020	Monthly NTA Statement 31 August 2020
3/09/2020	Share Purchase Plan - Extension of Closing Date
24/08/2020	Appendix 4E - Full Year Financial Statements 30 June 2020
20/08/2020	Receipt of shareholder notice
20/08/2020	Change in substantial holding
20/08/2020	Cleansing Notice - Placement
20/08/2020	Notice of Meeting
19/08/2020	Appendix 2A
17/08/2020	SPP - Offer Booklet Dispatch
17/08/2020	Share Purchase Plan - Cleansing Notice
14/08/2020	Receipt of shareholder notice
12/08/2020	Appendix 3B - SPP
12/08/2020	Proposed issue of Securities - CIE
12/08/2020	Share Purchase Plan
12/08/2020	Proposed New Investment Strategy
10/08/2020	Trading Halt
10/08/2020	Investment update and NTA statement 31 July 2020
8/07/2020	Investment update and NTA statement 30 June 2020

5. Information about CIE Shares

5.1. CIE capital structure

As at the date of this Bidder's Statement and according to the last ASIC 484 Form of CIE, there are 125,529,319 CIE Shares on issue.

5.2. Substantial holders

As at the date of this Bidder's Statement, according to the most recent substantial holder notices filed on the ASX, the substantial holders in CIE are:

Rank	Name	Shares	% of Shares
1	Wilson Asset Management Group	14,787,279	14.13
2	Lanyon Asset Management Pty Limited	10,701,396	8.86

5.3. CIE Share Price

The closing price of CIE Shares on the ASX on 25 September 2020 (being the last trading day that CIE Shares traded prior to the date of this Bidder's Statement) was \$0.635.

The VWAP of CIE Shares on the ASX for the one month before the date of this Bidder's Statement was \$0.632.

The VWAP of CIE Shares on the ASX for the five trading days before the date of this Bidder's Statement was \$0.625.

The highest recorded sale price of CIE Shares on the ASX in the four months before the date this Bidder's Statement was lodged with ASIC was \$0.685 on 9 June 2020.

The lowest recorded sale price of CIE Shares on the ASX in the four months before this Bidder's Statement was lodged with ASIC was \$0.575 on 27 May 2020.

During the four months prior to the date of the Offer, WAM has acquired CIE Shares under CIE's Placement to professional and sophisticated investors announced to the market on 12 August 2020 and through on-market purchases in September 2020. Under the Placement, WAM received 1,911,025 CIE Shares at the subscription price of \$0.625 per CIE Share, which were issued on 19 August 2020. WAM purchased 137,619 CIE Shares on-market during September 2020 at the price of \$0.625 per CIE Share.

5.4. WAM and its Associates' interests in CIE

As at the date of this Bidder's Statement, WAM had a Relevant Interest in 12.3% CIE Shares and its Associates held 1.5%, therefore WAM and its Associates had Voting Power of 13.8% in CIE.

6. Profile of the Enlarged Group

6.1. Introduction

The profile of the Enlarged Group will vary depending on the outcome of this Offer, and also the outcome of the CLF Offer that is currently open. The description of the Enlarged Group in this Section (unless otherwise specified) assumes that CIE will become a wholly-owned Subsidiary of WAM.

There is no guarantee of either or both of these outcomes occurring. If WAM does not acquire the 90% or more of CIE Shares required to entitle it to compulsorily acquire CIE Shares during the Offer Period, some of the benefits that would otherwise accrue to WAM if CIE were to become a wholly-owned Subsidiary of WAM may not be fully realised.

6.2. Overview of the Enlarged Group

The Enlarged Group would be publicly listed on the ASX. The Enlarged Group's investments will be managed by Wilson Asset Management, however CIE's investments will continue to be managed by the CIE Manager under the CIE Investment Management Agreement.

Assuming that WAM acquires 100% of CIE as part of the Offer, the Enlarged Group would have a combined pre-tax NTA of approximately \$1,339.2m as at 31 August 2020.

The estimated combined pre-tax NTA of the Enlarged Group does not include the impacts of the following:

- 4,752,220 new CIE Shares issued on 16 September 2020 under the SPP to existing CIE Shareholders at \$0.625 per CIE Share, raising \$3.0m; and
- acquisition of Concentrated Leaders Fund Limited by WAM through an off-market takeover offer scheduled to close on 19 October 2020 unless extended or withdrawn. As at 31 August 2020, CLF's pre-tax NTA was approximately \$65.5m.

On 3 September 2020, WAM lodged a bidder's statement and announced its intention to acquire all of the issued shares of Concentrated Leaders Fund Limited (ASX: CLF) by way of a conditional off-market takeover offer. The CLF Offer is currently open and will close on 19 October 2020 (unless extended or withdrawn). If the CLF Offer is successful, CLF will also become a wholly owned subsidiary of WAM and will form part of the Enlarged Group. See www.asx.com.au for details of the CLF Offer. The CLF Offer is not considered to be a material transaction for the purposes of this Bidder's Statement.

Under the CLF Offer, the maximum number of WAM Shares which would be issued as consideration is 28,290,524.

As at 31 August 2020	Pre-tax NTA \$'m	Adjustments \$'m	Total \$'m
WAM Capital Limited	1,258.7	(9.6)	1,249.1
Contango Income Generator Limited	90.1	-	90.1
Total Enlarged Group	1,348.8	(9.6)	1,339.2

Source:

- 1. WAM's pre-tax NTA is based on the announced August 2020 Investment Update.
- 2. CIE's pre-tax NTA is based on the Monthly NTA Statement 31 August 2020, as announced to the ASX and not adjusted for the Share Purchase Plan CIE Shares issued on 16 September 2020.
- 3. Adjustments to WAM's pre-tax NTA have been made to account for the existing ownership interest of WAM in CIE as at 31 August 2020. The existing interest in CIE, measured at its ASX last sale price as at 31 August 2020, has been eliminated and the CIE pre-tax NTA has been recognised in the Enlarged Group, assuming WAM acquires 100% of CIE.

6.3. Unaudited pro forma statements of financial position assuming WAM acquires 100% of CIE

Set out below is the unaudited pro forma statement of financial position of WAM, assuming that WAM acquires 100% of CIE as part of the Offer. The pro forma financial information is to be read in conjunction with the accompanying notes to and forming part of the pro forma statement of financial position as set out in this section. All amounts disclosed are presented in Australian dollars and are rounded to the nearest thousand dollars.

The pro forma financial information and adjustments of the Enlarged Group have been prepared on the basis that all CIE Shareholders accept the Offer and as if the proposed transaction had taken place on 30 June 2020:

			Pro Forma	
Unaudited Pro Forma Statement of Financial Position	Note	WAM Capital		Enlarged Group
as at 30 June 2020		\$	\$	\$
Current assets				
Cash and cash equivalents		203,400,924	-	203,400,924
Trade and other receivables		20,984,982	-	20,984,982
Financial assets	1	963,612,288	(8,705,059)	954,907,229
Financial assets - investment in controlled entity at FVTPL (CIE)	2	-	77,676,000	77,676,000
Current tax assets		22,676,280	-	22,676,280
Total current assets		1,210,674,474	68,970,941	1,279,645,415
Non-current assets				
Deferred tax assets		68,540,392		68,540,392
Total non-current assets		68,540,392	-	68,540,392
Total assets		1,279,214,866	68,970,941	1,348,185,807
Current liabilities				
Trade and other payables		32,857,926	-	32,857,926
Financial liabilities		897,913	-	897,913
Current tax liabilities		-	-	-
Total current liabilities		33,755,839	-	33,755,839
Non-current liabilities				
Deferred tax liabilities	3	23,778,042	4,014,410	27,792,452
Total non-current liabilities		23,778,042	4,014,410	27,792,452
Total liabilities		57,533,881	4,014,410	61,548,291
Net assets		1,221,680,985	64,956,531	1,286,637,516
Forth				
Equity	,	105000/404	FF F00 F31	1/00550705
Issued capital	4	1,352,964,191	55,589,574	1,408,553,765
Profits reserve	5	44,600,021	9,366,957	53,966,978
Accumulated losses		(175,883,227)		(175,883,227)
Total equity		1,221,680,985	64,956,531	1,286,637,516

The pro forma financial information of the Enlarged Group has not been adjusted for the following transactions and events for both WAM and CIE since 30 June 2020:

- 15,753,534 new CIE Shares issued on 19 August 2020 under the Placement to professional and sophisticated investors at \$0.625 per CIE Share, raising \$9.8m; and
- 4,752,220 new CIE Shares issued on 16 September 2020 under the SPP to existing CIE Shareholders at \$0.625 per CIE Share, raising \$3.0m; and
- acquisition of Concentrated Leaders Fund Limited by WAM through an off-market takeover offer scheduled to close on 19 October 2020 unless extended or withdrawn.
 As at 30 June 2020, for the purposes of the pro forma financial information of the Enlarged Group, CLF's pre-tax NTA was approximately \$71.5m.

On 3 September 2020, WAM lodged a bidder's statement and announced its intention to acquire all of the issued shares of Concentrated Leaders Fund Limited (ASX: CLF) by way of a conditional off-market takeover offer. The CLF Offer is currently open and will close on 19 October 2020 (unless extended or withdrawn). If the CLF Offer is successful, CLF will also become a wholly owned subsidiary of WAM and will form part of the Enlarged Group. See www.asx.com.au for details of the CLF Offer. The CLF Offer is not considered to be a material transaction for the purposes of this Bidder's Statement.

Under the CLF Offer, the maximum number of WAM Shares which would be issued as consideration is 28,290,524.

Note 1: Financial assets

	Pro Forma			
As at 30 June 2020	WAM Capital	Adjustments	Enlarged Group	
As at 30 Julie 2020	<u> </u>	Ÿ	•	
Financial assets	963,612,288	(8,705,059)	954,907,229	

Forming part of the pro forma adjustments to the 30 June 2020 statement of financial position for WAM is the elimination of the existing interest in CIE held at balance date in the financial assets of WAM. The existing interest in CIE as at 30 June 2020 is set out below:

WAM Capital's existing interest in CIE As at 30 June 2020

Total value of existing interest in CIE	8,705,059
ASX last sale price – as at 30 June 2020	\$0.65 / share
No. of shares held in CIE - as at 30 June 2020	13,392,399

Note 2: Financial assets - investment in controlled entity at FVTPL (CIE)

	Pro Forma		
	WAM Capital	Adjustments	Enlarged Group
As at 30 June 2020	\$	\$	\$
Financial assets – investment in controlled entity at FVTPL (CIE)	-	77,676,000	77,676,000

Forming part of the pro forma adjustments to the 30 June 2020 statement of financial position for WAM is the recognition of the investment in CIE, assuming WAM acquires 100% of CIE as part of the Offer. The value of CIE has been recognised utilising the audited pre-tax NTA of CIE as at 30 June 2020 (from the Annual Financial Report) for the purposes of the proforma financial information, set out below:

Value of investment in controlled entity at FVTPL

CIE pre-tax NTA	77,676,000
Add: Deferred tax liabilities	-
Less: Deferred tax assets	(1,467,000)
CIE 30 June 2020 Net Assets	79,143,000

Source: CIE's audited 2020 Annual Financial Report.

Note 3: Deferred tax liabilities

		Pro Forma	
	WAM Capital	Adjustments	Enlarged Group
As at 30 June 2020	\$	\$	\$
Deferred tax liabilities	23,778,042	4,014,410	27,792,452

In reference to the fair value gain on acquisition of a controlled entity as part of the pro forma adjustments (see Note 5 for further information), the tax effect of the transaction is recognised for the Enlarged Group as part of the recognition of the investment in CIE, assuming that WAM acquires 100% of CIE as part of the Offer.

Adjustments in relation to the tax effect of CIE joining WAM's tax consolidated group upon completion the Offer, assuming WAM acquires 100% of CIE, are not able to be determined at this time and have not been considered as part of the pro forma adjustments.

Note 4: Issued capital

As at 30 June 2020	WAM Capital	Adjustments \$	Enlarged Group \$
Issued capital	1,352,964,191	55,589,574	1,408,553,765

Forming part of the pro forma adjustments to the 30 June 2020 statement of financial position for WAM is the issue of WAM scrip to CIE shareholders in accordance with the terms of the Offer, as set out below:

Transaction details As at 30 June 2020

Offer: 1 WAM Share for every 3 CIE Shares	1 for 3
No. of CIE Shares on issue – as at 30 June 2020	105,023,565
No. of CIE Shares held by WAM – as at 30 June 2020	13,392,399
Remaining CIE Shares	91,631,166
No. of WAM Shares to be issued as Offer consideration	30,543,722
Scrip value (WAM Shares) - as at 30 June 2020	\$1.82 / share
Total Offer consideration value	55,589,574

Note 5: Profits reserve

	Pro Forma		
	WAM Capital	Adjustments	Enlarged Group
As at 30 June 2020	\$	\$	\$
Profits reserve	44,600,021	9,366,957	53,966,978

In recognising the acquisition of 100% of CIE in the Enlarged Group, the existing interest in CIE (measured at the ASX last sale price) is eliminated and the identifiable pre-tax NTA of CIE is recognised in consideration for the issue of WAM Shares as part of the Offer to CIE Shareholders.

The difference between the value of the WAM Shares issued and the existing interest in CIE, when compared to the identifiable pre-tax NTA of CIE is recognised through the statement of financial performance and resulting equity reserves.

Assuming the completion of the Offer as at 30 June 2020, the breakdown of the accounting recognition is set out below:

Transaction details As at 30 June 2020

Value of WAM Shares (see Note 4) – as at 30 June 2020	55.589.574
Value of With chares (see Note 4) as at 60 Julie 2020	30,303,374
Existing Relevant Interest in CIE, measured at the last sale price (see Note 1)	8,705,059
Total consideration	64,294,633
Identifiable pre-tax NTA of CIE (see Note 2) - as at 30 June 2020	77,676,000
Total fair value gain on acquisition of a controlled entity	13,381,367
Less: taxation expense, at 30%	4,014,410
Total fair value gain, after tax	9,366,957

6.4. Summary of information

The information included in this Section 6 is unaudited pro forma financial information for the Enlarged Group comprising of WAM and CIE as at 30 June 2020 on the basis that WAM acquires 100% of CIE, assuming completion of the Offer as at 30 June 2020. The pro forma financial information of the Enlarged Group has not been adjusted for the following transactions and events for both WAM and CIE since 30 June 2020:

- 15,753,534 new CIE Shares issued on 19 August 2020 under the Placement to professional and sophisticated investors at \$0.625 per CIE Share, raising \$9.8m; and
- 4,752,220 new CIE Shares issued on 16 September 2020 under the SPP to existing CIE Shareholders at \$0.625 per CIE Share, raising \$3.0m; and
- acquisition of Concentrated Leaders Fund Limited by WAM through an off-market takeover offer scheduled to close on 19 October 2020 unless extended or withdrawn.
 As at 30 June 2020, for the purposes of the pro forma financial information of the Enlarged Group, CLF's pre-tax NTA was approximately \$71.5m.

On 3 September 2020, WAM lodged a bidder's statement and announced its intention to acquire all of the issued shares of Concentrated Leaders Fund Limited (ASX: CLF) by way of a conditional off-market takeover offer. The CLF Offer is currently open and will close on 19 October 2020 (unless extended or withdrawn). If the CLF Offer is successful, CLF will also become a wholly owned subsidiary of WAM and will form part of the Enlarged Group. See www.asx.com.au for details of the CLF Offer. The CLF Offer is not considered to be a material transaction for the purposes of this Bidder's Statement.

Under the CLF Offer, the maximum number of WAM Shares which would be issued as consideration is 28,290,524.

The unaudited pro forma combined statements of financial position of the Enlarged Group as at 30 June 2020 (Pro Forma Statement of Financial Position) presented above has been produced with reference to the audited financial report for WAM as at 30 June 2020 and the audited financial report for CIE as at that date. The financial information reflects the latest publicly available information for CIE that is capable of being compared.

The WAM Board has assessed the requirements of AASB 2013-5 Amendments to Australian Accounting Standards – Investment Entities and have applied the criteria set out in that standard to the operations of WAM. WAM is therefore considered to be an investment entity and as a result, the investment in CIE (being a wholly owned entity of the WAM) would not be consolidated into the financial statements, but rather is accounted for as a financial asset at FVTPL. After acquisition, subsequent movements in the fair value of the identifiable net assets of the controlled entity are recoded as fair value gains or losses through the statement of financial performance.

The pro forma financial information does not represent what the Enlarged Group would look like on a combined basis, since it is not possible to produce this information from publicly available information and due to the accounting policies of WAM. No adjustments for potential synergy benefits have been included as the exact timing and amount of those benefits cannot be reliably estimated. However, the Pro Forma Statement of Financial Position does reflect the issue of new WAM Shares to CIE Shareholders, and other adjustments required as a result of this Offer.

The Pro Forma Statement of Financial Position is for illustrative purposes only and is based on numerous assumptions that may or may not reflect the actual financial position of the Enlarged Group after completion of the Offer. In addition, the Pro Forma Statement of Financial Position is presented in a summary format and therefore does not contain all the disclosures required under the Corporations Act.

Financial information relating to CIE has been sourced from its publicly available information. WAM has relied on the information in the monthly investment portfolio update and NTA reports of CIE as well as audited financial reports issued by CIE to prepare the financial information contained in this document.

WAM does not, except as required by law, make any representation or warranty, express or implied, as to the accuracy or completeness of this information. WAM has not been able to independently verify any of the financial information relating to CIE used in this document, for providing pro forma financial information.

The pro forma financial information presented in this section should also be read in conjunction with the risks set out in Section 8, other information contained in this Bidder's Statement, and the accounting policies of WAM and CIE as disclosed in their most recent audited financial reports.

The accounting policies of WAM and CIE are not considered materially different. The significant accounting policies adopted in the preparation of the historical financial information are disclosed in WAM's audited Annual Report for the year ended 30 June 2020.

6.5. No forward looking financial statements

This document does not provide revenue or profit projections for the Enlarged Group. WAM does not believe that it has reasonable grounds to include revenue or profit forecasts in this Bidder's Statement. WAM believes that the inclusion of revenue or profit forecasts would be unduly speculative and potentially misleading for CIE Shareholders.

6.6. Prospective financial information of the Enlarged Group

WAM has given careful consideration as to whether a reasonable basis exists to produce reliable and meaningful forecast financial information on the Enlarged Group. The WAM Board has concluded that providing forecast financial information would be misleading. A reasonable basis does not exist for producing forecasts that would be sufficiently meaningful and reliable, particularly considering the effect that variations in key inputs may have on future earnings performance.

The Enlarged Group will continue to pursue investment activities – primarily investing in equities listed on the ASX – to achieve its stated objectives.

The future performance of the Enlarged Group is dependent on the performance of its investments. In turn, the performance of these investments is impacted by investee company-specific factors and prevailing industry conditions. In addition, a range of external factors including economic growth rates, interest rates, exchange rates and macro-economic conditions impact the overall equity market and these investments. As such, the WAM Board do not believe it is possible or appropriate to accurately predict the future performance of the Enlarged Group's investments and, therefore, it's performance.

7. WAM's intentions

7.1. General

This Section 7 sets out WAM's intentions in relation to:

- a) the continuation of CIE's business;
- b) any significant changes to be made to CIE's business, including any redeployment of CIE's fixed assets;
- c) the future employment of CIE's employees (where applicable); and
- d) potential outcomes from the WAM EGM.

WAM's intentions are based on the information known, and the circumstances that exist, at the date of this Bidder's Statement.

WAM has reviewed information in the public domain concerning CIE, its business and the general business environment in which it operates. Based on its review of that material, and on its current assessment of CIE's operations, WAM's intentions in relation to CIE are summarised below. Statements set out in this Section are statements of current intention only which may change as new information becomes available or circumstances change.

7.2. Strategic Review

As WAM does not currently have access to all material information, facts and circumstances which it considers necessary to assess the operational, commercial, taxation and financial implications of its current intentions, final decisions on these matters have not been made. Upon successful completion of the Offer, and in the event that WAM controls more than 50% of CIE's Shares, WAM proposes to conduct a review of the strategy, operations, activities, assets and employees (based on publicly available information CIE has no fixed assets or employees) of CIE in light of the information which then becomes available to it (Strategic Review).

This Strategic Review will:

- a) confirm whether WAM should proceed with a realisation of CIE's investments and conduct a return of capital;
- b) explore other investment strategies for CIE's remaining cash; and
- c) seek to make appropriate provisions for any of CIE's ongoing liabilities.

Final decisions will only be reached in light of all material facts and circumstances. Accordingly, the statements set out in this Section 7 are statements of current intention only which may change as new information becomes available or circumstances change.

7.3. WAM's intentions upon controlling 90% or more of CIE Shares

This Section sets out WAM's intentions if it controls 90% or more of CIE Shares and becomes entitled to proceed to compulsory acquisition of CIE Shares under the Corporations Act.

- a) WAM intends to undertake the Strategic Review;
- b) WAM currently intends to proceed to compulsory acquisition of the remaining CIE Shares if it becomes entitled to do so under the Corporations Act;
- c) WAM will seek to remove CIE from the Official List of the ASX;
- d) WAM intends to replace all CIE Directors with its nominees;
- e) WAM intends to work constructively with the CIE Manager;
- f) Based on publicly available information, it appears that CIE has no employees. If at the relevant time CIE does employ any person, then, subject to the Strategic Review, it may be necessary to redeploy or make some of CIE's employees redundant. If redundancies do occur, the relevant employees will receive benefits in accordance with their contractual and other legal entitlements.

7.4. WAM's intentions upon controlling more than 50.1% but less than 90% of CIE Shares

If WAM, by virtue of acceptance of the Offer, acquires a Relevant Interest in CIE of more than 50.1% but less than 90%, its current intentions are as set out below.

- a) CIE will become a controlled entity of WAM;
- b) WAM intends to undertake the Strategic Review;
- c) Subject to the Corporations Act and the Strategic Review, WAM proposes to seek the appointment of a majority of WAM nominees to CIE's Board. WAM has, in its EGM requisition, proposed Geoff Wilson AO and Glenn Burge as candidates for CIE's Board although it has not made any other decisions as to who would be nominated for appointment. WAM may also seek to add to, replace or reorganise the roles of a proportion of the members of the CIE Board to reflect the proportionate ownership interest of WAM and other CIE Shareholders;
- d) WAM intends to work constructively with the CIE Manager;
- e) WAM may seek to remove CIE from the Official List of the ASX;
- f) WAM may, at some later time, acquire further CIE Shares in a manner consistent with the Corporations Act;
- g) If WAM becomes entitled at some later time to exercise general compulsory acquisition rights under Part 6A.2 of the Corporations Act, it may exercise those rights; and
- h) WAM may consider acquiring additional CIE Shares under the "creep" provisions of Item 9 in section 611 of the Corporations Act (which allows for WAM and its Associates to acquire up to 3% of CIE Shares every 6 months);
- i) WAM intends to undertake the activities referred to in this Section 7.4 to the extent permitted by its control of CIE.

7.5. WAM's intentions upon controlling less than 50.1% of CIE Shares

WAM reserves its right to declare the Offer free from the 50.1% minimum acceptance Condition (or any other Condition) to the Offer. However, WAM has not decided whether it will free the Offer of the Conditions.

In that circumstance:

- a) WAM has a Relevant Interest in less than 50.1%, to the extent possible through its non-controlling holding of CIE Shares, WAM will endeavour to give effect to the intentions set out in Section 7.4. However, WAM considers that its ability to affect the intentions set out in Section 7.4 will be significantly limited if WAM is unable to achieve the level of shareholding.
- b) WAM may consider acquiring additional CIE Shares under the "creep" provisions of Item 9 in section 611 of the Corporations Act (which allows for WAM and its Associates to acquire up to 3% of CIE Shares every 6 months) until it achieves a majority Voting Power. WAM has not yet decided whether it will acquire further CIE Shares in the event it controls less than 50.1%.

7.6. Limitations on intentions

The intentions and statements of future conduct set out in this Section 7 must be read as being subject to:

- a) the outcome of the Strategic Review;
- b) the law (including the Corporations Act) and the Listing Rules, including in particular the requirements of the Corporations Act and the Listing Rules in relation to conflicts of interest and "related party" transactions given that, if WAM gains control of CIE but does not acquire all of the CIE Shares, it will be treated as a related party of CIE for these purposes; and
- c) the legal obligations of the CIE Directors at the time, including any nominees of WAM, to act in good faith and in the best interests of CIE and for proper purposes and to have regard to the interest of all CIE Shareholders (in which regard the role of independent Directors of CIE will also be important).

7.7. WAM EGM outcomes

On 11 September 2020, the WAM Entities lodged a notice under section 249D of the Corporations Act, requesting that CIE call and arrange the WAM EGM at which WAM would seek to put a resolution to members to remove Mark Kerr and Don Clarke as CIE Directors, and appoint Geoff Wilson AO and Glenn Burge in their place, with a view to subsequently appointing Wilson Asset Management as CIE's new investment manager. Wilson Asset Management is also the manager of WAM.

CIE has not yet called the WAM EGM and has until 2 October 2020 to do so. Depending on the date CIE sets for the WAM EGM, the Offer may have already closed, in which case WAM may have already moved to the actions described above in this Section 7.

If the WAM EGM occurs during the Offer Period, CIE Shareholders will have the opportunity to vote on resolutions to appoint Geoff Wilson AO and Glenn Burge to the CIE Board before the Offer closes.

8. Risk factors

8.1. Introduction

The future investment performance of WAM and the future performance of the WAM Shares may be influenced by a range of factors, many of which are outside the control of WAM. WAM is subject to risks that are both of a general nature and risks which are specific to its business activities.

This Section describes what WAM considers to be some of the key risks associated with WAM's business, the industry in which it operates, and the risks associated with an investment in WAM.

The risks listed in this Section should not be considered to be an exhaustive list of every possible risk associated with accepting the Offer, the industry in which WAM operates, or the risks associated with an investment in WAM now or in the future. The types of risks to which WAM is exposed can change over time and vary with changes in economic, technological, environmental and regulatory conditions. The occurrence or consequences of some of these risks are partially or completely outside of the control of WAM.

The selection of risks is based on an assessment of a combination of the probability of the risk occurring, the ability to mitigate the risk and the impact of the risk if it did occur. They are not listed in order of likelihood of occurrence or impact. The assessment is based on the knowledge of WAM's Directors and management as at the date of this Bidder's Statement, but there is no guarantee or assurance that the importance of these or other risks will not change or that no other new risks will emerge. Any of these risks, and any other risks that may emerge, may have a material adverse effect on the operations, performance and position of WAM. There can be no guarantee that WAM will achieve its stated objectives or that any forward-looking statements or forecasts will eventuate.

Before accepting the Offer, you should read the entire Bidder's Statement carefully and satisfy yourself that you have a sufficient understanding of the risks described in this Section 8, together with the other information set out in this Bidder's Statement. You should also consider whether accepting the Offer is suitable for you in light of your investment objectives, financial situation, taxation position and particular needs. If you do not understand any part of this Bidder's Statement or have any questions about whether to accept the Offer, you should consult your accountant, financial adviser, stockbroker, solicitor or other independent and qualified professional adviser before deciding whether to accept the Offer.

8.2. Risks specific to the Offer

The following risk factors may arise as a result of the Offer and the acquisition of CIE Shares by WAM.

Conditions to the Offer

As described in Schedule 1 of this Bidder's Statement, the Offer is subject to a number of conditions.

Subject to any statutory withdrawal rights that may be available, CIE Shareholders who accept the Offer while it still remains conditional will no longer be able to trade their CIE Shares on the ASX or withdraw their acceptance of the Offer. CIE Shareholders should be aware that the market price of CIE Shares may exceed the implied price under the Offer during the Offer Period.

For those CIE Shareholders who accept the Offer and the conditions remain unsatisfied at the end of the Offer Period, and are not otherwise waived by WAM, there is no obligation on WAM to issue WAM Shares to you as consideration for your CIE Shares. In those circumstances, any acceptances of the Offer will be void and you would then be free to deal with your CIE Shares.

Limited withdrawal rights

CIE Shareholders may only withdraw their acceptance of the Offer in limited circumstances. Otherwise, CIE Shareholders will be unable to withdraw their acceptances, even if the value of the WAM Shares to be issued to CIE Shareholders as consideration varies from the date of their acceptance.

If the Offer is extended, and does not become unconditional, you may not receive your WAM Shares in time to receive the WAM Dividend. If the Offer does not become unconditional and is extended for less than one month, you will not be able to withdraw your acceptance of the Offer, and you will not receive the WAM Dividend.

Issue of WAM Shares as the Offer Consideration

CIE Shareholders are being offered consideration under the Offer that comprises a specified number of WAM Shares, rather than a number of WAM Shares with a specified market value. As a result, the value of the consideration will fluctuate depending on the value of WAM Shares. Furthermore, if current WAM Shareholders or current CIE Shareholders who accept the Offer do not wish to hold shares in the Enlarged Group and seek to sell their shares, this may have material adverse effect on the Enlarged Group and the market price of WAM Shares.

Accordingly, the market value of WAM Shares at the time you receive them may vary significantly from their market value on the date of your acceptance of the Offer. This may result in the value of the consideration to CIE Shareholders increasing as well as decreasing.

Less than 100% ownership

It is possible that WAM may not acquire all CIE Shares under the Offer. The impact upon WAM of acquiring less than 100% will depend upon the final level of ownership acquired. However, the existence of minority shareholders in CIE could have an impact on WAM's intentions regarding CIE (refer to Section 7).

CIE information

In preparing the information on CIE included in this Bidder's Statement, WAM has relied on publicly available information of CIE only. WAM has not had direct access to CIE Directors or its auditors. In addition, it is possible that additional risks may exist in relation to CIE's operations that are unknown to WAM.

There is also a risk that the publicly available financial information of CIE and utilised by WAM in formulating the Offer and preparing this Bidder's Statement is not materially correct.

Reporting risks

WAM has assumed that CIE has met all of its reporting obligations and that disclosures made by it are true and are not misleading in any material respects.

Dilution risks

As the consideration payable for the Offer consists of WAM Shares, the interests of existing WAM Shareholders (which may include CIE Shareholders following completion of the Offer) will be diluted to some degree, with respect to the percentage ownership of the Company.

Taxation risks

Tax liabilities are the responsibility of each individual investor, and WAM is not responsible either for taxation or penalties incurred by CIE Shareholders. CIE Shareholders should consult their own taxation advisers to ascertain the tax implications of their investment.

Taxation considerations

The effects of taxation can be complex and may change over time. A summary of the current Australian income tax implications associated with acceptance of the Offer is outlined in Section 10. However, this summary is general in nature and, as the circumstances for each CIE Shareholder may vary, CIE Shareholders should seek professional taxation advice in relation to their own position.

This list is not exhaustive and CIE Shareholders should read this Bidder's Statement in full. If CIE Shareholders require further information on these risks, they should seek professional advice.

Potential impact on subsequent bids

WAM's shareholding may become of such a size that a third party would not be able to successfully make a takeover bid for CIE without the support of WAM.

8.3. Risks specific to WAM

Manager risk

The risk that WAM's investment objectives will not be achieved and/or it may underperform the S&P/ASX All Ordinaries Accumulation Index (WAM's benchmark) or may underperform other listed investment companies in the same asset class. The achievement of WAM's objectives (including the ability to pay dividends) depends on a range of factors, including its portfolio selection, the skills of the WAM's investment team and the actual performance of investments.

Risks on WAM's investments

WAM may invest in entities with a limited trading history or small capital base. These investments may pose greater risk than investments in larger capitalised entities or entities in other market sectors. While the development of sound risk management procedures and sound due diligence can minimise the risks of investment in such companies, the risks are material and significant.

Listed Investment Company risk

A listed investment company's share price may trade at a premium or discount to net tangible asset backing per share. The operations and financial performance of listed investment companies are driven by a number of factors, including, but not limited to:

- domestic and international economic growth;
- performance of domestic and international financial markets;
- domestic and international political events; and
- relevant regulation in Australia and overseas.

Business risk

The profitability and success of WAM is highly dependent on the ability of its Directors and the investment manager to assess business risks and make appropriate investment decisions. No guarantee can be given in respect of the future earnings of WAM or the earnings and capital appreciation of its investments.

8.4. Risks specific to CIE

Manager risk

The risk that CIE's investment objective will not be achieved and/or it may underperform CIE's benchmark or may underperform other listed investment companies in the same asset class. The achievement of CIE's objective (including the ability to pay dividends) depends on a range of factors, including its portfolio selection, the skills of the CIE's investment manager's employees, the timing of implementation of protection strategies, and actual performance of investments.

Risks on CIE's investments

The change to CIE's investment strategy exposes CIE to risk associated with the WCM Quality Global Long Short Strategy, through an investment in the WCM Focused Global Growth Long-Short Fund Limited – CIE has not provided information to CIE Shareholders in relation to the risks associated with this strategy. WCM Focused Global Growth Long-Short Fund Limited is managed by WCM Investment Management.

Through an investment in the WCM Focused Global Growth Long-Short Fund Limited, CIE's underlying investments may include, but are not limited to, publicly traded or privately issued or negotiated common stocks, preferred stocks, stock warrants and rights, corporate and U.S. government debt, bonds, notes or other debentures or debt participations, ETFs, currency hedges, convertible securities, fixed income securities, swaps (including total return swaps), and options (purchased or written).

CIE's investment strategy may expose CIE Shareholders to global entities with a limited trading history or small capital base, as well as exposing CIE's portfolio to risks associated with taking short equity positions and the financial instruments used.

A global investment strategy also introduces foreign exchange risk to CIE's portfolio. These investments may pose greater risk than investments in domestic entities. While the development of sound risk management procedures and sound due diligence can minimise the risks of investment in such companies, the risks are material and significant.

Any derivative financial instruments implemented under the new investment strategy will be linked to the underlying value of physical assets and are used by WCM Investment Management as an alternative to investing in physical assets because of their cost and liquidity efficiency or, to protect the value of the underlying physical assets from adverse price movements over time. Gains or losses can result from investments in derivatives Investments in derivatives may increase the losses that would otherwise have been experienced had the derivatives not been used.

There is a risk that the use of derivatives can have a negative impact due to an adverse movement in the underlying asset or where the position is difficult or costly to reverse or maintain. The use of derivatives potentially exposes CIE to counterparty, legal and documentation risks. In addition to any risk associated with the underlying asset for which a derivative is valued, derivative prices are affected by other factors including market liquidity, interest rates, and counterparty risk.

Listed Investment Company risk

A listed investment company's share price may trade at a premium or discount to net tangible asset backing per share. The operations and financial performance of listed investment companies are driven by a number of factors, including, but not limited to:

- domestic and international economic growth;
- performance of domestic and international financial markets;
- domestic and international political events; and
- relevant regulation in Australia and overseas.

Business risk

The profitability and success of CIE is highly dependent on the ability of its Directors and investment manager to assess business risks and make appropriate investment decisions. No guarantee can be given in respect of the future earnings of CIE or the earnings and capital appreciation of its investments.

8.5. General risks

Regulation and Publicity

Changes to the regulatory environment for listed investment companies, the Corporations Act, ASIC policy, Listing Rules and their commercial application and/or publicity may have an impact on CIE and WAM's investments and may affect the value of CIE Shares and WAM Shares.

Changes in Accounting Policy

WAM and CIE must report and prepare financial statements in accordance with prevailing accounting standards and policies. There may be changes in these accounting standards and policies in the future which may have an adverse impact on WAM and CIE (as applicable) reported financial performance and financial position.

Economic conditions

The price at which WAM Shares and CIE Shares trade on the ASX may be determined by a range of factors including movements in local and international equity and bond markets, general investor sentiment in those markets, inflation, interest rates, general economic conditions and outlooks.

Sharemarket conditions

The market for WAM Shares WAM Shares and CIE Shares may also be affected by a wide variety of events and factors including COVID-19, variations in WAM and CIE's respective operating results, recommendations by securities analysts, and the operating and trading price performance of other listed entities that investors consider to be comparable to WAM or CIE (as applicable). Some of these factors could affect WAM or CIE's share price regardless of underlying operating performance.

Currency Risk

Shareholders who reside outside of Australia, or who rely on funding denominated in a currency other than the Australian dollar, should be aware of the impact that fluctuations in exchange rates may have on the value of their investments in, and returns.

Forward looking statements

Certain information in this Bidder's Statement constitutes forward looking information that is subject to risks and uncertainties and a number of assumptions, which may cause the actual expenditure and achievement of milestones and other outcomes of the combined group to be different from the expectations expressed or implied in this Bidder's Statement.

8.6. Risks relating to the Enlarged Group

Implementation Risks

Irrespective of the outcome of the Offer, the Investment Management Agreement between CIE and CIE Manager is for a fixed term. While WAM intends to work constructively with CIE Manager there is no guarantee that CIE Manager will carry out WAM's intentions or instructions in relation to the management of CIE's portfolio of investments.

Diversification

As holders of WAM Shares, following implementation of the Offer, CIE Shareholders will be exposed to a more diversified portfolio of investments than CIE. That diversification may continue in the future. A more diverse portfolio of investments may mean CIE Shareholders are exposed to a more diverse range of risks.

WAM Shares

Under the Offer, CIE Shareholders are issued with WAM Shares in consideration for the transfer of CIE Shares. The share price performance of a listed company, such as WAM is subject to general fluctuations in the share market. These fluctuations can result from a wide variety of factors, some of which are beyond the control of WAM such as:

- general economic conditions, including changes in inflation rates, interest rates and exchange rates;
- variations in the domestic and international markets for listed stocks;
- changes in government policy, legislation or regulation; and
- inclusion or removal from market indices.

In particular, the share prices in any companies may reflect a diverse range of issues which are not specific to the company concerned, such as acts of terrorism and the general state of the economy. These market fluctuations may adversely affect the price of WAM Shares. An investment in WAM Shares therefore carries risks that the share price may fall and no guarantee can be made about the declaration of dividends or returns of capital.

9. Sources of Consideration

9.1. Offer Consideration

The consideration for the acquisition of CIE Shares under the Offer is to be satisfied by the issue of WAM Shares. The Offer Consideration is 1 WAM Share for every 3 CIE Shares.

The total number of WAM Shares to be issued will depend on the number of CIE Shares that WAM acquires under the Offer.

The calculations in this section on the maximum consideration to be paid by WAM assume that, as a result of the Offer, WAM acquires all CIE Shares on issue at the date of this Bidder's Statement that WAM does not already own.

9.2. Maximum number of WAM Shares

The maximum number of WAM Shares which are required to be issued under the Offer if acceptances are received for all the CIE Shares on issue at the date of this Bidder's Statement that WAM does not already own is approximately 36,696,092 (excluding rounding adjustments).

WAM Shares issued under the Offer are exempt from ASX Listing Rule 7.1 (which prohibits issues exceeding 15% of a company's share capital in any 12-month period). Accordingly, WAM has the capacity to issue the maximum number of WAM Shares that may be required under the Offer.

10. Tax considerations

10.1. Introduction

The following is a summary of the potential Australian income tax and capital gains tax (CGT) consequences generally applicable to CIE Shareholders who dispose of CIE Shares under the Offer. All CIE Shareholders are advised to seek independent professional advice about their particular circumstances and non-resident CIE Shareholders should seek their own advice on the Australian and foreign taxation consequences of any sale of their CIE Shares. The information in this section is based upon the Australian law and administrative practice in effect at the date of this Bidder's Statement, but is general in nature and is not intended to be an authoritative or complete statement of the laws applicable to the particular circumstances of a CIE Shareholder.

In particular this summary is only applicable to CIE Shareholders that are Australian residents for income tax purposes and hold their CIE Shares on capital account for investment purposes and only considers the Australian tax position.

This summary does not apply to CIE Shareholders:

- who hold their CIE Shares for the purpose of speculation or a business of dealing in securities (e.g. as trading stock or as revenue assets);
- who acquired their CIE Shares pursuant to an employee share scheme operated by CIE;
- who are subject to taxation of financial arrangement rules in Division 230 of the Income Tax Assessment Act 1997 in relation to gains or losses on their CIE Shares;
- that may be subject to special tax rules, such as financial institutions, insurance companies, partnerships (except where expressly stated), tax exempt organisations, trusts (except where expressly stated), superannuation funds (except where expressly stated) or temporary residents; or
- are non-residents of Australia who hold their CIE Shares in carrying on a business through a permanent establishment in Australia.

CIE Shareholders who are tax residents of a country other than Australia (whether or not they are also residents, or are temporary residents, of Australia for tax purposes) should take into account the tax consequences of the Offer under the laws of their country of residence, as well as under Australian law.

10.2. CGT consequences on the disposal of CIE Shares

A CIE Shareholder that accepts the Offer and whose CIE Shares are subsequently transferred to WAM, is taken to have disposed of their CIE Shares for Australian CGT purposes. Shareholders make a capital gain on the disposal of their CIE Shares to the extent that the market value of the Offer consideration exceeds the tax cost base of the CIE Shareholders' CIE Shares (prior to the availability of the CGT discount (see below) and any losses available to be offset against the capital gain).

The capital proceeds received for the disposal of the CIE Shares should be calculated by reference to the market value of WAM Shares that each CIE Shareholder will receive.

The sum of all capital gains derived by a CIE Shareholder in the year in which the transaction occurs, reduced by any capital loss incurred during that year, or carried forward from prior years (known as the net capital gain), should be included in the assessable income of the CIE Shareholder.

Alternatively, a CIE Shareholder will make a capital loss on the transfer of their shares to WAM, equal to the amount by which the reduced cost base of their CIE Shares is more than the capital proceeds (i.e. value of shares in WAM) they receive under the Offer. A capital loss may be used to offset against a capital gain made in the same income year or be carried forward to offset a capital gain made in a future income year (subject to the satisfaction of certain loss recoupment tests which apply if the CIE Shareholder is a company or trust).

10.3. Cost base of CIE Shares generally

If a CIE Shareholder does not dispose of their CIE Shares under the Offer and their CIE Shares are compulsorily acquired in accordance with Part 6A.1 of the Corporations Act, those CIE Shareholders will also be treated as having disposed of their CIE Shares for CGT purposes.

In this case, the date of the CGT event will be the date when the CIE Shareholder ceases to be the owner of the CIE Shares.

10.4. CGT rollover

Where WAM acquires 80% or more of the Shares in CIE, a CIE Shareholder may be eligible to choose for scrip for scrip CGT rollover relief under Subdivision 124-M of the Income Tax Assessment Act 1997 (Cth) in respect of the disposal of CIE Shares, so that any CGT payable on the disposal of their CIE Shares is deferred.

If a CIE Shareholder elects for CGT rollover relief to apply, any capital gain that they would otherwise make on the disposal of their CIE Shares will be disregarded, as the only consideration that CIE Shareholders will receive in respect of their CIE Shares are replacement shares in WAM.

The benefit of choosing scrip for scrip rollover relief will depend on the individual circumstances of each CIE Shareholder and therefore CIE Shareholders should discuss this with their tax advisers.

Where rollover relief is chosen, the tax cost base of the WAM Shares received by the CIE Shareholders should equal the tax cost base of the CIE Shares disposed of. This tax cost base will be allocated on a proportionate basis across the WAM Shares received. This cost base will be relevant in determining any future CGT liability on the subsequent disposal of the WAM Shares.

For the purposes of determining whether the CGT discount applies on any subsequent disposal of the WAM Shares, the date on which the CIE Shareholders will be deemed to have acquired the WAM Shares will be the day they acquired their original CIE Shares.

Generally, where CIE Shareholders prepare their income tax return on the basis that scrip for scrip rollover has been applied, this will be sufficient evidence of making the choice to obtain rollover relief.

In certain circumstances, it may be necessary for WAM to make a joint election with CIE Shareholders in order for CGT rollover relief to apply. To the extent a joint election is necessary for rollover relief to apply then, at the request of the CIE Shareholder and provided the CIE Shareholder has informed WAM in writing of the cost base of their CIE Shares, WAM may make the joint election. WAM confirms that it will not make any choice or election under the CGT rollover provisions that would prevent CIE Shareholders from obtaining CGT rollover relief where available.

However, if the WAM Offer becomes unconditional and WAM does not increase its holding to at least 80% of the CIE Shares on issue as a result of the Offer, CGT rollover relief will not be available for any CIE Shareholders.

10.5. CGT discount

Any CIE Shareholder who is an individual, the trustee of a trust or a complying superannuation entity that does not choose CGT rollover relief may be entitled to a 'CGT discount' in calculating the net capital gain to be included in their assessable income.

- CIE Shareholders will be eligible for CGT discount provided that:
- the CIE Shares were acquired at least 12 months before disposal to WAM;
- the CIE Shareholder did not choose to index the cost base of their CIE Shares; and
- the CGT discount is applied to the capital gain after any available capital losses are first applied to reduce the capital gain.

The CGT discount allows a CIE Shareholder who is an individual or the trustee of a trust to discount the capital gain by 50%, including only the remaining 50% of the capital gain in the taxable income of that individual or trust.

A CIE Shareholder that is a complying superannuation entity may discount the capital gain by 33.3% and include 66.6% of the capital gain in the taxable income of that complying superannuation entity.

The CGT discount is not available to a CIE Shareholder that is a company.

10.6. Stamp duty and GST

Neither the sale of CIE Shares, nor the issue of WAM Shares should give rise to any stamp duty liability for existing CIE Shareholders.

10.7. Goods and Services Tax (GST)

The sale of CIE Shares by existing shareholders as contemplated will not attract GST. Similarly, no GST will be payable on the acquisition of WAM Shares.

Where shareholders are registered or required to be registered for GST, any GST incurred on expenses that relate to the sale of existing shares or acquisition of new shares may not be recoverable if the individual shareholder exceeds the financial acquisitions threshold as set out in the relevant GST legislation. However, a reduced input tax credit equal to 75% of the GST incurred may still be available if the acquisition constitutes a reduced credit acquisition.

Where CIE Shareholders are not registered, or required to be registered for GST, no GST implications should arise in relation to the Offer.

10.8. Obtain your own taxation advice

Do not rely on the comments or the statements contained in this Bidder's Statement as advice about your own affairs. The taxation laws are complex and there could be implications in addition to those generally described in this Bidder's Statement.

Accordingly, consult your own tax advisers for advice applicable to your individual needs and circumstances. To the extent permitted by law, WAM does not accept any responsibility for tax implications for individual CIE Shareholders.

WAM is not a registered tax agent under the Tax Agent Services Act 2009 (Cth) and cannot provide tax advice to specific CIE Shareholders. CIE Shareholders should obtain tax advice from a registered tax agent on the consequences of accepting the Offer.

11. Additional information

11.1. WAM is a disclosing entity

As WAM is offering WAM Shares as consideration for the acquisition of CIE Shares, the Corporations Act requires that this Bidder's Statement includes all information that would be required for a prospectus for an offer of WAM Shares under sections 710 to 713 of the Corporations Act. WAM does not need to issue a separate prospectus for the Offer of the WAM Shares as the offer occurs under a takeover bid.

WAM is a disclosing entity (as defined in section 111AC of the Corporations Act) for the purposes of section 713 of the Corporations Act that is subject to regular reporting and disclosure obligations. Specifically, like all listed companies, WAM is required to continuously disclose to the market any information it has, which a reasonable person would expect to have a material effect on the price or the value of WAM's Shares.

WAM Shares have been quoted on the ASX during the 12 months before the date of this Bidder's Statement. For this reason, WAM is only required to disclose information in this Bidder's Statement that would usually be required where its shares have been continuously quoted securities.

In general terms, where WAM's shares are continuously quoted securities, the prospectus is only required to contain information on the effect of the Offer on WAM and the rights and liabilities attaching to the WAM Shares. It is not necessary to include general information on all of the assets and liabilities, financial position and performance, profits and losses or prospects of the issuing company unless that information has been excluded from a continuous disclosure notice in accordance with the Listing Rules and it is information that investors and their professional advisers would reasonably require for the purpose of making an informed assessment of those matters and the rights and liabilities attaching to the WAM Shares.

Information that is already in the public domain is not reported in this Bidder's Statement other than that which is considered necessary to make this Bidder's Statement complete. WAM, as a disclosing entity under the Corporations Act, states that:

- a) it is subject to regular reporting and disclosure obligations;
- copies of documents lodged with ASIC by WAM (not being documents referred to in section 1274(2)(a) of the Corporations Act) may be obtained from, or inspected at, an ASIC office; and
- c) it will provide a copy of each of the following documents, free of charge, to any person on request during the Offer Period:
 - i) the financial statements of WAM for the 12 months ended 30 June 2020 (being the annual report most recently lodged by WAM with ASIC);
 - ii) the half-year financial report lodged for the six months ended 31 December 2019 lodged with ASIC by WAM; and
 - iii) all continuous disclosure notices given by WAM after the lodgement of that annual financial report with ASIC and before the lodgement of this Bidder's Statement with ASIC.

If you would like to receive a copy of any of these documents or publications, please call Chairman Geoff Wilson AO on 02 9247 6755 or our Chief Financial Officer Jesse Hamilton on 0401 944 807.

Copies of all documents lodged with ASIC in relation to WAM can be inspected at the registered office of WAM during normal office hours. A list of announcements made by WAM on the ASX between 30 June 2020 and the date of this Bidder's Statement as it appears in Schedule 2.

Other than information contained in this Bidder's Statement, there is no information which has been excluded from a continuous disclosure notice in accordance with the Listing Rules and is information that a CIE Shareholder or a professional adviser to a CIE Shareholder would reasonably require for the purposes of making an informed assessment of:

- a) the assets and liabilities, financial position and performance, profits and prospects of WAM; or
- b) the rights and liabilities attaching to WAM Shares.

None of the information referred to in this Section 11.1 is incorporated by reference into this Bidder's Statement or is included with this Bidder's Statement.

11.2. Capital Structure of WAM

The total number of WAM Shares on issue is 725,465,456 listed shares. There are no other WAM securities on issue.

On 3 September 2020, WAM lodged a bidder's statement and announced its intention to acquire all of the issued shares of Concentrated Leaders Fund Limited (ASX: CLF) by way of a conditional off-market takeover offer. The CLF Offer is currently open and will close on 19 October 2020 (unless extended or withdrawn). If the CLF Offer is successful, CLF will also become a wholly owned subsidiary of WAM and will form part of the Enlarged Group. See www.asx.com.au for details of the CLF Offer. The CLF Offer is not considered to be a material transaction for the purposes of this Bidder's Statement.

Under the CLF Offer, the maximum number of WAM Shares which would be issued as consideration is 28,290,524.

11.3. Rights and liabilities attaching to WAM Shares

The rights and liabilities attaching to ownership of the WAM Shares arise from a combination of WAM's constitution, statute, the Listing Rules and general law.

A summary of the significant rights and liabilities attaching to WAM's Shares is set out below. This summary is not exhaustive, nor does it constitute a definitive statement of the rights and liabilities of WAM Shareholders.

Voting at a general meeting

Subject to any shares which may in the future be issued with special or preferential rights, every member present in person at a general meeting of WAM or by proxy, representative or attorney has one vote on a show of hands and, on a poll, one vote for each fully paid

WAM Share held. On a poll, partly paid shares confer a fraction of a vote in proportion to the amount paid up on the share.

Meeting of members

Each WAM Shareholder is entitled to receive notice of, attend and vote at meetings of WAM and to receive all notices, accounts and other documents required to be sent to WAM Shareholders under WAM's constitution, the Corporations Act and the Listing Rules.

Dividends

The WAM Directors may authorise the payment of dividends to members as and when permitted by the Corporations Act. Dividends are payable on all WAM Shares in proportion to the amount of the total issue price paid or credited as paid for the WAM Shares. This is subject to any special or preferential rights attached to any class of shares created after the issue of the WAM Shares. If a fully paid share is on issue for only part of the period in respect of which a dividend is payable, the dividend for that share will be apportioned and paid according to the proportion that the part of that period for which the share is on issue and is fully paid is of the total period for which the dividend is payable.

Transfer of WAM Shares

WAM Shares may be transferred by a proper transfer effected under the ASX Settlement Operating Rules, or by a written instrument of transfer in any usual form or in any other form approved by the WAM Directors.

The Directors may only decline to register a transfer of WAM Shares (other than a proper transfer under the ASX Settlement Operating Rules) where permitted or required to do so under the Listing Rules. If the Directors refuse to register a transfer, WAM must, within five Business Days after the transfer is lodged with WAM, give the party lodging the transfer written notice of the refusal and the reason for refusal. The WAM Directors must decline to register a transfer of WAM Shares when required by the Corporations Act or the Listing Rules.

Issue of further WAM Shares

The Directors may allot, issue, grant options in respect of, or otherwise dispose of, further shares in WAM on terms as they see fit. However, the WAM Directors must act in accordance with the restrictions imposed by WAM's constitution, the Corporations Act, the Listing Rules and any rights for the time being attached to the shares in any special class of those shares.

Winding-up

If WAM is wound up, the liquidator may, with the sanction of a special resolution, divide among the WAM Shareholders the whole or any part of the property of WAM and may for that purpose set such value as the liquidator considers fair upon any property to be so divided and may determine how the divisions is to be carried out as between the WAM Shareholders or different classes of WAM Shareholders.

Share buy-back

Subject to the provisions of the Corporations Act and the Listing Rules, WAM may buy back WAM Shares on terms and at times determined by the WAM Directors.

Variation of class rights

Unless otherwise provided by WAM's constitution or by the terms of issue of a class of shares, the rights attaching to any class of shares may be varied or abrogated:

- a) with the consent in writing of the holders of three-quarters of the issued shares included in that class; or
- b) with the sanction of a special resolution passed at a separate meeting of the holders of those shares.

In either case, the holders of not less than 10% of the votes in the class of shares, the rights of which have been varied or abrogated, may apply to a court of competent jurisdiction to exercise its discretion to set aside that variation or abrogation.

Dividend reinvestment plan

WAM's constitution authorises the WAM Directors to establish and maintain dividend reinvestment plans (whereby any member may elect that dividends payable by WAM be reinvested by way of subscription for shares in WAM).

Indemnities

To the extent permitted by law, WAM indemnifies every officer and former officer of WAM against a liability incurred by the person acting in that capacity. In addition, every officer is indemnified out of the assets of WAM against any liability incurred in defending any civil or criminal proceedings in which judgment is given in favour of the officer, or in which the officer is acquitted.

Alteration of WAM's constitution

WAM's constitution can only be amended by special resolution passed by at least 75% of WAM Shareholders present and voting at a general meeting of WAM. WAM must give at least 28 days written notice of its intention to propose a resolution as a special resolution.

11.4. Capital structure of CIE

- a) WAM's Offer is to acquire all of the issued CIE Shares that it does not already control.
- b) At the date of this Bidder's Statement, there are 125,529,319 CIE Shares on issue.

11.5. Details of WAM's Relevant Interests in CIE Shares

At the date of this Bidder's Statement, the number of CIE Shares in which WAM had a Relevant Interest is set out below:

Class of securities	Total number in class	% of Shares on issue
CIE Ordinary Shares	15,441,043	12.3

As at the date of this Bidder's Statement WAM and its Associates have Voting Power of 13.8% in CIE.

WAM reserves the right to purchase CIE Shares outside the Offer at any time during the Offer Period, subject to the Corporations Act.

11.6.Details of WAM's Directors Relevant Interests in CIE Shares

At the date of this Bidder's Statement, no WAM Director has any Relevant Interests in CIE Shares.

11.7. Valuation of CIE Shares for the purpose of the minimum bid price rule

WAM may rely on the provisions of ASIC Corporations (Minimum Bid Price) Instrument 2015/1068. That Instrument provides relief from the requirement in section 621(4) of the Corporations Act and allows a bidder up to five Business Days before the date of the Offer to value quoted Securities that it is offering as part of its bid consideration, for the purpose of determining whether the value of WAM's bid consideration at least equals the maximum value of consideration paid by WAM for CIE Shares in the four months prior to the date of this Bidder's Statement (described in Section 5.3).

Relying on that relief, WAM may value WAM Shares to be provided under the Offer at any date nominated by WAM (Valuation Date) up to five Business Days before the date of the Offer. The valuation is determined by calculating the VWAP of WAM Shares in the ordinary course of trading on the ASX during two full trading days before the Valuation Date.

Under section 621(4A) (as inserted in the Corporations Act by ASIC Corporations (Minimum Bid Price) Instrument 2015/1068) the following information is provided:

- a) valuation of WAM Shares under section 621(4A): \$2.290 per WAM Share
- b) Valuation Date: 8 October 2020
- c) trading days on which valuation is based: 6 October 2020 and 7 October 2020

The value of the Offer Consideration per CIE Share for the purposes of section 621 of the Corporations Act using this valuation is therefore A\$0.763.

The value exceeds the maximum value of consideration paid by WAM for CIE Shares in the four months prior to the date of the Offer (as described in Section 5.3). This value exceeds the maximum value of the consideration per CIE Share paid by WAM in the four months prior to the date of the bid, as set out in below and in Section 5.3.

During the four months prior to the date of the Offer, WAM has acquired CIE Shares under CIE's Placement to professional and sophisticated investors announced to the market on 12 August 2020 and through on-market purchases in September 2020. Under the Placement, WAM received 1,911,025 CIE Shares at the subscription price of \$0.625 per CIE Share, which were issued on 19 August 2020. WAM purchased 137,619 CIE Shares on-market during September 2020 at the price of \$0.625 per CIE Share.

WAM Shares can fall as well as rise and may be subject to varied and unpredictable influences. Accordingly, there is no guarantee that WAM Shares will continue to be worth equal to or above \$2.290. The highest and lowest closing prices of WAM Shares on the ASX in the four months before the Valuation Date were \$2.31 and \$1.77 respectively.

11.8. No collateral benefits

Neither WAM nor any of its Associates has, during the period of four months ending on the day immediately before the date of the Offer, given, or offered or agreed to give, a benefit to another person likely to induce the other person, or an Associate, to:

- a) accept the Offer; or
- b) dispose of CIE Shares,

which benefit was not offered to all CIE Shareholders under the Offer.

11.9.No escalation agreements

Neither WAM nor any Associate of WAM has entered into any escalation agreement that is prohibited by section 622 of the Corporations Act.

11.10. Regulatory approvals

There are no regulatory approvals that WAM needs to obtain before acquiring CIE Shares before or under the Offer.

11.11. ASIC modifications and exemptions

ASIC has published various class orders and instruments providing for modifications and exemptions that apply generally to all persons, including WAM, about the operation of Chapter 6 of the Corporations Act. WAM may rely on this relief.

WAM has relied on the modification to section 636(3) of the Corporations Act in paragraph 5(d) of ASIC Class Order 13/521 'Takeover Bids' to include references to certain statements in documents lodged with ASIC and ASX without obtaining the consent of the persons to whom those statements are attributed.

WAM will make available a copy of the documents containing these statements (or relevant extracts from these documents), free of charge to CIE Shareholders who request them before the end of the Offer Period.

To obtain a copy of these documents (or the relevant extracts) CIE Shareholders should contact WAM by calling Chairman Geoff Wilson AO on 02 9247 6755 or Chief Financial Officer Jesse Hamilton on 0401 944 807.

11.12. Foreign Shareholders

If you are a Foreign Shareholder, unless WAM otherwise determines that:

- a) it is lawful and not unduly onerous and not unduly impracticable to make the Offer to you and to issue you with WAM Shares on acceptance of the Offer; and
- b) it is not unlawful for you to accept the Offer by the law of the relevant place outside Australia, its or external territories,

you will not be entitled to receive WAM Shares as consideration for your CIE Shares.

Instead, WAM will:

- a) arrange for the issue to the Nominee of the number of WAM Shares to which you and all other Foreign Shareholders would have been entitled but for clause 3.1 of Schedule 1;
 and
- cause the Nominee to offer for sale those WAM Shares within five Business Days after the end of the Offer Period in the manner, at the price and on other terms as are determined by the Nominee; and
- c) pay to you the amount ascertained under the formula in clause 7.3 of Schedule 1.

Notwithstanding anything else contained in this Bidder's Statement, WAM is not under any obligation to spend any money or undertake any action to satisfy itself of the eligibility of Foreign Shareholders to receive WAM Shares.

11.13. Consents

PwC Australia has given (and before lodgement of this Bidder's Statement has not withdrawn) its written consent to be named as the Original Australian legal adviser to the Offer in the form and context in which it is named.

Mills Oakley has given (and before lodgement of this Bidder's Statement has not withdrawn) its written consent to be named as the Principal Australian legal adviser to the Offer in the form and context in which it is named.

Boardroom Pty Limited (ACN 003 209 836) (Boardroom) has given and before lodgement of this Bidder's Statement has not withdrawn its written consent to be named as registry for the Offer in the form and context in which it is named.

Each of PwC Australia, Mills Oakley and Boardroom:

- a) has not authorised or caused the issue of this Bidder's Statement;
- b) does not make, or purport to make, any statement in this Bidder's Statement other than as specified in this Section; and
- c) to the maximum extent permitted by law, expressly disclaims all liability in respect of, makes no representation regarding and takes no responsibility for, any part of this Bidder's Statement other than the reference to its name and the statements (if any) included in this Bidder's Statement with the consent of that party as specified in this Section.

11.14. Date for determining holders of CIE Shares

For the purposes of section 633 of the Corporations Act, the date for determining the people to whom information is to be sent under items 6 and 12 of section 633(1) is 30 September 2020.

11.15. Fees and benefits payable to Directors and advisers

Other than as set out below or elsewhere in this Bidder's Statement, no:

- a) WAM Director or proposed WAM Director;
- b) person named in this Bidder's Statement as performing a function in a professional, advisory or other capacity in connection with the preparation or distribution of this Bidder's Statement; or
- c) promoter of WAM,

has, or had within two years before the date of this Bidder's Statement, any interest in:

- a) the formation or promotion of WAM;
- any property acquired or proposed to be acquired by WAM in connection with its formation or promotion or in connection with the issue of WAM Shares under the Offer; or
- c) the offer of WAM Shares under this Bidder's Statement,

and no amounts have been paid or agreed to be paid and no benefits have been given or agreed to be given to any Director or a proposed Director of WAM as an inducement to become, or to qualify as, a Director of WAM or to any of those persons otherwise for services rendered by them in connection with the formation or promotion of WAM or the offer of WAM Shares under this Bidder's Statement.

The WAM Directors have the following Relevant Interests in WAM Shares (either held directly, held by entities controlled by them or held by entities of which they are WAM Directors or held by family members) at the date of this Bidder's Statement:

Director	Balance at 30 June 2020	Balance as at date of this Bidder's Statement
Geoff Wilson AO	412,884	412,884
Dr Philippa Ryan	-	-
James Chirnside	35,815	35,815
Lindsay Mann	63,880	63,880
Kate Thorley	57,491	57,491
Matthew Pancino	-	-
TOTAL	570,070	570,070

Insurance and indemnity arrangements are in place for WAM Directors and officers of WAM. To the extent permitted by law, WAM indemnifies every person who is or has been an officer against:

- a) any liability to any person (other than WAM, related entities or a major shareholder) incurred while acting in that capacity and good faith; and
- b) costs and expenses incurred by that person in that capacity in successfully defending legal proceedings and ancillary matters.

For this purpose 'officer' means any WAM Director or secretary of WAM or any Subsidiary of WAM. Under the terms of the indemnity, WAM must maintain indemnity insurance coverage for a period of seven years after the officer held their position in WAM and allow the officer access to certain records after ceasing to hold office in WAM for a period of seven years.

PwC Australia and Mills Oakley have acted as Australian legal advisers to WAM in relation to the Offer. WAM estimates that it will pay amounts totalling \$85,000 in respect of this work.

11.16. Expiry date

No WAM Shares will be issued on the basis of the Offer contained in this Bidder's Statement after the date that is 13 months after the date of this Bidder's Statement.

11.17. Other material information

There is no other information material to the making of a decision about whether or not to accept the Offer (being information that is known to WAM and has not previously been disclosed to the holders of CIE Shares) other than as disclosed in this Bidder's Statement.

11.18. Costs and expenses

All costs and expenses of the preparation and circulation of this Bidder's Statement and the Offer will be paid by WAM.

12. Definitions and interpretations

12.1. Definitions

The following defined terms are used throughout this Bidder's Statement unless the context otherwise requires:

\$ or c	Australian dollars or cents.	
Acceptance Form	The acceptance form for the Offer accompanying this Bidder's Statement.	
Announcement Date	Means 28 September 2020, the date on which this Bidder's Statement was lodged with ASIC.	
ASIC	Australian Securities and Investments Commission.	
Associate	Has the same meaning given to that term in section 9 of the Corporations Act.	
ASX	ASX Limited (ACN 008 624 691) on the market it administers being Australian Securities Exchange (as the context requires).	
ASX Settlement	ASX Settlement Pty Limited (ACN 008 504 532).	
ASX Settlement Operating Rules	The operating rules of the settlement facility provided by ASX Settlement.	
Bidder's Statement	This document, being the statement made by WAM under Part 6.5 Division 2 of the Corporations Act.	
Board	Means the board of directors of CIE.	
Broker	A person who is a stockbroker and participant in CHESS.	
Business Day	Monday to Friday inclusive, except New Year's Day, Good Friday, Easter Monday, Christmas Day, Boxing Day and any other day that the ASX declares is not a business day.	
CGT	Capital gains tax.	
CHESS	The Clearing House Electronic Subregister System which provides for the electronic transfer, settlement and registration of securities in Australia.	
CHESS Holding	A holding of CIE Shares on the CHESS subregister of CIE.	
CIE	Contango Income Generator Limited (ACN 160 959 991)	
CIE DRP	Means the CIE dividend reinvestment plan.	
CIE Manager	Contango Asset Management Limited (ACN 080 277 998).	
CIE Share	A fully paid ordinary share in the capital of CIE.	
CIE Shareholder	A registered holder of CIE Shares.	
Condition	Each condition set out in Schedule 1.	
Controlling Participant	Has the meaning given in the ASX Settlement Operating Rules.	
Corporations Act	The Corporations Act 2001 (Cth).	
Defeating Condition	Each condition set out in clause 8.1 of Schedule 1.	
Director	Means a director of CIE.	

Issuer Sponsored Holding A holding of CIE Shares on CIE's issuer sponsored sub-register. The listing rules of the ASX. Means the nominee to be appointed by WAM for the purposes detailed in Section 11.12. NTA Net tangible assets. Offer The offer as set out in Schedule 1 of this Bidder's Statement itself and includes a reference to that offer as varied in accordance with the Corporations Act. Offer Consideration Consideration offered by WAM for CIE Shares as set out in Section 9. Offer Period The period for which the Offer remains open. Offers The Offer and each of the other offers made in the same terms for CIE Shares and includes a reference to those offers as varied in accordance with the Corporations Act. Official List The official list of entities that the ASX has admitted and not removed. Official Quotation Official quotation of a Security on a market operated by the ASX. Placement Means CIE's placement to professional and sophisticated investors, announced to the market on 12 August 2020. Prescribed Occurrence Each event referred to in Sections 652C(1) and (2) of the Corporations Act which is set out in section 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date 30 September 2020. Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. Rights All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions, and all rights to receive any dividends or other distributions, or to receive or subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.			
outside Australia and its external territories, to whom it is unlawful for WAM to make the Offer or for whom it is unlawful to accept the Offer. FVTPL Fair value through profit and loss. Governmental Agency Any government, semi-government, administrative, fiscal, judicial or regulatory body, department, commission, authority, tribunal, agency or entity. GST Goods and services tax. Investment Management Agreement between CIE and the CIE Manager dated on or around 21 September 2020. Issuer Sponsored Holding A holding of CIE Shares on CIE's issuer sponsored sub-register. Listing Rules The listing rules of the ASX. Means the nominee to be appointed by WAM for the purposes detailed in Section 11.12. NTA Net tangible assets. Offer The offer as set out in Schedule 1 of this Bidder's Statement itself and includes a reference to that offer as varied in accordance with the Corporations Act. Offer Period The period for which the Offer remains open. Offers The Offer and each of the other offers made in the same terms for CIE Shares and includes a reference to those offers as varied in accordance with the Corporations Act. Official List The Official quotation of a Security on a market operated by the ASX. Placement Means CIE's placement to professional and sophisticated investors, announced to the market on 12 August 2020. Prescribed Occurrence Each event referred to in Sections 652C(1) and (2) of the Corporations Act. which is set out in section 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive on subscribe for shares, stock units, notes, bonds, options or other distributions and all rights to receive any dividends or other distributions, or to receive on subscribe for shares, stock units, notes,	Enlarged Group	WAM with CIE as a wholly owned Subsidiary.	
Any government, semi-government, administrative, fiscal, judicial or regulatory body, department, commission, authority, tribunal, agency or entity. GST Goods and services tax. Investment Management Investment Management Agreement between CIE and the CIE Manager dated on or around 21 September 2020. Issuer Sponsored Holding A holding of CIE Shares on CIE's issuer sponsored sub-register. Listing Rules The listing rules of the ASX. Nominee Means the nominee to be appointed by WAM for the purposes detailed in Section 11.12. NTA Net tangible assets. Offer The offer as set out in Schedule 1 of this Bidder's Statement itself and includes a reference to that offer as varied in accordance with the Corporations Act. Offer Consideration Consideration offered by WAM for CIE Shares as set out in Section 9. Offers The Offer and each of the other offers made in the same terms for CIE Shares and includes a reference to those offers as varied in accordance with the Corporations Act. Official List The Official list of entities that the ASX has admitted and not removed. Official Quotation Official quotation of a Security on a market operated by the ASX. Placement Means CIE's placement to professional and sophisticated investors, announced to the market on 12 August 2020. Prescribed Occurrence Each event referred to in Sections 652C(1) and (2) of the Corporations Act which is set out in section 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date 30 September 2020. Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. Rights All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive or subscribe for shares, stock units, notes, bonds, options or other receive or subscribe for shares, stock units, notes, bonds, options or other rec	Foreign Shareholder	outside Australia and its external territories, to whom it is unlawful for	
regulatory body, department, commission, authority, tribunal, agency or entity. GST Goods and services tax. Investment Management Investment Agreement between CIE and the CIE Manager dated on or around 21 September 2020. Issuer Sponsored Holding A holding of CIE Shares on CIE's issuer sponsored sub-register. Listing Rules The listing rules of the ASX. Nominee Means the nominee to be appointed by WAM for the purposes detailed in Section 11.12. NTA Net tangible assets. Offer The offer as set out in Schedule 1 of this Bidder's Statement itself and includes a reference to that offer as varied in accordance with the Corporations Act. Offer Consideration Consideration offered by WAM for CIE Shares as set out in Section 9. Offers The offer and each of the other offers made in the same terms for CIE Shares and includes a reference to those offers as varied in accordance with the Corporations Act. Official List The official list of entities that the ASX has admitted and not removed. Official Quotation Official quotation of a Security on a market operated by the ASX. Placement Means CIE's placement to professional and sophisticated investors, announced to the market on 12 August 2020. Prescribed Occurrence Each event referred to in Sections 652C(1) and (2) of the Corporations Act which is set out in section 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date 30 September 2020. Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. Rights All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive on sylvidends or other distributions or ot or receive on subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	FVTPL	Fair value through profit and loss.	
Investment Management dated on or around 21 September 2020. Issuer Sponsored Holding A holding of CIE Shares on CIE's issuer sponsored sub-register. Listing Rules The listing rules of the ASX. Nominee Means the nominee to be appointed by WAM for the purposes detailed in Section 11.12. NTA Net tangible assets. Offer The offer as set out in Schedule 1 of this Bidder's Statement itself and includes a reference to that offer as varied in accordance with the Corporations Act. Offer Consideration Consideration offered by WAM for CIE Shares as set out in Section 9. Offers The offer and each of the other offers made in the same terms for CIE Shares and includes a reference to those offers as varied in accordance with the Corporations Act. Official List The official list of entities that the ASX has admitted and not removed. Official Quotation Official quotation of a Security on a market operated by the ASX. Placement Means CIE's placement to professional and sophisticated investors, announced to the market on 12 August 2020. Prescribed Occurrence Each event referred to in Sections 652C(1) and (2) of the Corporations Act which is set out in section 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date 30 September 2020. Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive on subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	Governmental Agency	regulatory body, department, commission, authority, tribunal, agency or	
Agreement dated on or around 21 September 2020. Issuer Sponsored Holding A holding of CIE Shares on CIE's issuer sponsored sub-register. Listing Rules The listing rules of the ASX. Nominee Means the nominee to be appointed by WAM for the purposes detailed in Section 11.12. NTA Net tangible assets. Offer The offer as set out in Schedule 1 of this Bidder's Statement itself and includes a reference to that offer as varied in accordance with the Corporations Act. Offer Consideration Consideration offered by WAM for CIE Shares as set out in Section 9. Offers The offer and each of the other offers made in the same terms for CIE Shares and includes a reference to those offers as varied in accordance with the Corporations Act. Official List The official list of entities that the ASX has admitted and not removed. Official Quotation Official quotation of a Security on a market operated by the ASX. Placement Means CIE's placement to professional and sophisticated investors, announced to the market on 12 August 2020. Prescribed Occurrence Each event referred to in Sections 652C(1) and (2) of the Corporations Act which is set out in section 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date 30 September 2020. Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. Rights All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive o subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	GST	Goods and services tax.	
Listing Rules The listing rules of the ASX. Means the nominee to be appointed by WAM for the purposes detailed in Section 11.12. NTA Net tangible assets. Offer The offer as set out in Schedule 1 of this Bidder's Statement itself and includes a reference to that offer as varied in accordance with the Corporations Act. Offer Consideration Consideration offered by WAM for CIE Shares as set out in Section 9. Offer Period The period for which the Offer remains open. Offers The Offer and each of the other offers made in the same terms for CIE Shares and includes a reference to those offers as varied in accordance with the Corporations Act. Official List The official list of entities that the ASX has admitted and not removed. Official Quotation Official quotation of a Security on a market operated by the ASX. Placement Means CIE's placement to professional and sophisticated investors, announced to the market on 12 August 2020. Prescribed Occurrence Each event referred to in Sections 652C(1) and (2) of the Corporations Act which is set out in section 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date 30 September 2020. Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. Rights All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive o subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	Investment Management Agreement		
Means the nominee to be appointed by WAM for the purposes detailed in Section 11.12. NTA Net tangible assets. Offer The offer as set out in Schedule 1 of this Bidder's Statement itself and includes a reference to that offer as varied in accordance with the Corporations Act. Offer Consideration Consideration offered by WAM for CIE Shares as set out in Section 9. Offer Period The period for which the Offer remains open. Offers The Offer and each of the other offers made in the same terms for CIE Shares and includes a reference to those offers as varied in accordance with the Corporations Act. Official List The official list of entities that the ASX has admitted and not removed. Official Quotation Official quotation of a Security on a market operated by the ASX. Placement Means CIE's placement to professional and sophisticated investors, announced to the market on 12 August 2020. Prescribed Occurrence Each event referred to in Sections 652C(1) and (2) of the Corporations Act which is set out in sections 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date 30 September 2020. Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. Rights All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive o subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	Issuer Sponsored Holding	A holding of CIE Shares on CIE's issuer sponsored sub-register.	
Nort A Net tangible assets. Offer The offer as set out in Schedule 1 of this Bidder's Statement itself and includes a reference to that offer as varied in accordance with the Corporations Act. Offer Consideration Consideration offered by WAM for CIE Shares as set out in Section 9. Offer Period The period for which the Offer remains open. Offers The Offer and each of the other offers made in the same terms for CIE Shares and includes a reference to those offers as varied in accordance with the Corporations Act. Official List The official list of entities that the ASX has admitted and not removed. Official Quotation Official quotation of a Security on a market operated by the ASX. Placement Means CIE's placement to professional and sophisticated investors, announced to the market on 12 August 2020. Prescribed Occurrence Each event referred to in Sections 652C(1) and (2) of the Corporations Act which is set out in section 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date 30 September 2020. Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive or subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	Listing Rules	The listing rules of the ASX.	
The offer as set out in Schedule 1 of this Bidder's Statement itself and includes a reference to that offer as varied in accordance with the Corporations Act. Offer Consideration Consideration offered by WAM for CIE Shares as set out in Section 9. Offer Period The period for which the Offer remains open. Offers The Offer and each of the other offers made in the same terms for CIE Shares and includes a reference to those offers as varied in accordance with the Corporations Act. Official List The official list of entities that the ASX has admitted and not removed. Official Quotation Official quotation of a Security on a market operated by the ASX. Placement Means CIE's placement to professional and sophisticated investors, announced to the market on 12 August 2020. Prescribed Occurrence Each event referred to in Sections 652C(1) and (2) of the Corporations Act which is set out in section 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date 30 September 2020. Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. Rights All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive o subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	Nominee		
includes a reference to that offer as varied in accordance with the Corporations Act. Offer Consideration Consideration offered by WAM for CIE Shares as set out in Section 9. Offer Period The period for which the Offer remains open. Offers The Offer and each of the other offers made in the same terms for CIE Shares and includes a reference to those offers as varied in accordance with the Corporations Act. Official List The official list of entities that the ASX has admitted and not removed. Official Quotation Official quotation of a Security on a market operated by the ASX. Placement Means CIE's placement to professional and sophisticated investors, announced to the market on 12 August 2020. Prescribed Occurrence Each event referred to in Sections 652C(1) and (2) of the Corporations Act which is set out in section 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date 30 September 2020. Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. Rights All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive o subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	NTA	Net tangible assets.	
Offer Period The period for which the Offer remains open. The Offer and each of the other offers made in the same terms for CIE Shares and includes a reference to those offers as varied in accordance with the Corporations Act. Official List The official list of entities that the ASX has admitted and not removed. Official Quotation Official quotation of a Security on a market operated by the ASX. Placement Means CIE's placement to professional and sophisticated investors, announced to the market on 12 August 2020. Prescribed Occurrence Each event referred to in Sections 652C(1) and (2) of the Corporations Act which is set out in section 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date 30 September 2020. Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive o subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	Offer	includes a reference to that offer as varied in accordance with the	
The Offer and each of the other offers made in the same terms for CIE Shares and includes a reference to those offers as varied in accordance with the Corporations Act. Official List The official list of entities that the ASX has admitted and not removed. Official Quotation Official quotation of a Security on a market operated by the ASX. Placement Means CIE's placement to professional and sophisticated investors, announced to the market on 12 August 2020. Prescribed Occurrence Each event referred to in Sections 652C(1) and (2) of the Corporations Act which is set out in section 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date 30 September 2020. Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive o subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	Offer Consideration	Consideration offered by WAM for CIE Shares as set out in Section 9.	
Shares and includes a reference to those offers as varied in accordance with the Corporations Act. Official List The official list of entities that the ASX has admitted and not removed. Official Quotation Official quotation of a Security on a market operated by the ASX. Placement Means CIE's placement to professional and sophisticated investors, announced to the market on 12 August 2020. Prescribed Occurrence Each event referred to in Sections 652C(1) and (2) of the Corporations Act which is set out in section 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date 30 September 2020. Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. Rights All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive o subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	Offer Period	The period for which the Offer remains open.	
Official Quotation Official quotation of a Security on a market operated by the ASX. Placement Means CIE's placement to professional and sophisticated investors, announced to the market on 12 August 2020. Prescribed Occurrence Each event referred to in Sections 652C(1) and (2) of the Corporations Act which is set out in section 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date 30 September 2020. Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. Rights All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive o subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	Offers	Shares and includes a reference to those offers as varied in accordance with	
Placement Means CIE's placement to professional and sophisticated investors, announced to the market on 12 August 2020. Prescribed Occurrence Each event referred to in Sections 652C(1) and (2) of the Corporations Act which is set out in section 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date 30 September 2020. Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive o subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	Official List	The official list of entities that the ASX has admitted and not removed.	
announced to the market on 12 August 2020. Prescribed Occurrence Each event referred to in Sections 652C(1) and (2) of the Corporations Act which is set out in section 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date 30 September 2020. Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive o subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	Official Quotation	Official quotation of a Security on a market operated by the ASX.	
which is set out in section 8.1(c) of Schedule 1 in this Bidder's Statement. Register Date 30 September 2020. Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. Rights All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive o subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	Placement		
Relevant Interest Has the meaning given in sections 608 and 609 of the Corporations Act. All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive o subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	Prescribed Occurrence		
Rights All accretions, rights or benefits of whatever kind attaching to or arising from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive o subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	Register Date	30 September 2020.	
from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive o subscribe for shares, stock units, notes, bonds, options or other securities, declared, paid or issued by CIE or any of its Subsidiaries.	Relevant Interest	Has the meaning given in sections 608 and 609 of the Corporations Act.	
Security Has the meaning as given in section 92 of the Corporations Act.	Rights	from CIE Shares directly or indirectly after the date of this Bidder's Statement, including, without limitation, all dividends or other distributions and all rights to receive any dividends or other distributions, or to receive or subscribe for shares, stock units, notes, bonds, options or other securities,	
	Security	Has the meaning as given in section 92 of the Corporations Act.	

Security Interest	Any:
Security interest	 a) "security interest" as defined in the Personal Properties Securities Act Law;
	 b) security for payment of money, performance of obligations or protection against default (including a mortgage, bill of sale, charge, lien, pledge, trust, power or title retention arrangement, right of setoff, assignment of income, garnishee order or monetary claim and flawed deposit arrangements); c) a thing or preferential interest or arrangement of any kind giving a person priority or preference over claims of other persons or creditors with respect to any property or asset, and includes any agreement to create any of them or allow them to exist.
SPP	Means CIE's share purchase plan, announced to the market on 12 August 2020.
Strategic Review	Means the detailed strategic review proposed by WAM detailed in Section 7.2 of this Bidder's Statement.
Subsidiary	Has the meaning given in the Corporations Act.
Takeovers Panel	The Australian Government Takeovers Panel.
Voting Power	Has the same meaning given to that term in section 610 of the Corporations Act.
VWAP	Volume-weighted average price.
WAM	WAM Capital Limited (ACN 086 587 395).
WAM Board	The board of directors of WAM.
WAM Director	A director of WAM.
WAM Dividend	A 7.75 cents per WAM Share fully franked dividend announced by WAM on 8 July 2020.
WAM Share	A fully paid ordinary share in the capital of WAM.
WCM Investment Management	WCM Investment Management, LLC.
Wilson Asset Management	Wilson Asset Management (International) Pty Limited (ACN 081 047 118).

12.2. Interpretation

In this Bidder's Statement, unless the context requires otherwise:

- c) a reference to a word includes the singular and the plural of the word and vice versa;
- d) a reference to a person in this Bidder's Statement or any other document or agreement includes its successors and permitted assigns;
- e) a reference to a gender includes any gender;
- f) a reference to an item in a Section, Schedule, Annexure or Appendix is a reference to an item in the section of or schedule, annexure or appendix to this Bidder's Statement and references to this Bidder's Statement include its schedules and any annexures;
- g) if a word or phrase is defined, then other parts of speech and grammatical forms of that word or phrase have a corresponding meaning;
- h) a term which refers to a natural person includes a company, a partnership, an association, a corporation, a body corporate, a joint venture or a Governmental Agency;
- i) headings are included for convenience only and do not affect interpretation;
- j) a reference to a document or agreement including this Bidder's Statement, includes a reference to that document or agreement as amended, novated, supplemented, varied or replaced from time to time;
- k) a reference to a thing includes a part of that thing and includes but is not limited to a right;
- I) the terms included, including and similar expressions when introducing a list of items do not exclude a reference to other items of the same class or genus;
- m) a reference to a statute or statutory provision includes but is not limited to:
 - i) a statute or statutory provision which amends, extends, consolidates or replaces the statute or statutory provision;
 - ii) a statute or statutory provision which has been amended, extended, consolidated or replaced by the statute or statutory provision; and
 - iii) subordinate legislation made under the statute or statutory provision including but not limited to an order, regulation, or instrument;
- n) reference to \$, A\$, Australian Dollars or dollars, c or cents is a reference to the lawful tender for the time being and from time to time of the Commonwealth of Australia; and
- o) a reference to an asset includes all property or title of any nature including but not limited to a business, a right, a revenue and a benefit, whether beneficial, legal or otherwise.

13. Approval of Bidder's Statement

This Bidder's Statement has been approved by a unanimous resolution of the WAM Directors of WAM.

Dated 28 September 2020

Signed for and on behalf of WAM Capital Limited.

Geoff Wilson AO

Chairman

Schedule I – Terms of the Offer

1. Offer

- 1.1 WAM offers, to each holder of CIE Shares at 7.00pm (AEST) on the Register Date, to acquire all of their CIE Shares together with all the Rights attaching to them on the terms set out in this Schedule 1. This Offer extends to any person who is able during the Offer Period to give good title to CIE Shares and has not already accepted an Offer for those CIE Shares.
- 1.2 This Offer extends to all CIE Shares that WAM does not have a Relevant Interest in that:
 - (a) exist at the Register Date; and
 - (b) are issued during the period from the Register Date to the end of the Offer Period due to the conversion of, or exercise of the rights attached to, any securities which confer on the holder the right to have CIE Shares issued to them.
- 1.3 By accepting this Offer, you undertake to transfer to WAM not only the CIE Shares to which the Offer relates but also all the Rights attached to the CIE Shares.
- 1.4 WAM Shares issued under this Offer will rank equally in all respects with WAM Shares on issue at the Register Date.
- 1.5 This Offer is dated 13 October 2020.

2. Offer Consideration

2.1 Subject to clause 3.1 of this Schedule 1, the price under the Offer Consideration is 1 WAM Share for every 3 CIE Shares you own.

3. Foreign Shareholders

3.1 If at the time this Offer is made to you, you are a Foreign Shareholder and WAM does not make a determination under Section 11.12, you will not be entitled to receive WAM Shares as consideration for your CIE Shares. Instead, you will receive a cash amount calculated under clause 7.3 of this Schedule 1.

4. Fractional entitlements

4.1 If you accept the Offer and are entitled to receive a fraction of a WAM Share, the number of WAM Shares you are entitled to under this Offer is rounded up to the next whole number of WAM Shares if it is 0.5 or more, and rounded down to the nearest whole number if it is less than 0.5. If you are entitled to a fraction of a cent, that fraction is rounded to the nearest cent on the same basis. If WAM reasonably believes that you have manipulated your holding of CIE Shares to take advantage of

rounding up, then any fractional entitlement is aggregated or rounded down to the next whole number of WAM Shares.

5. Offer Period

5.1 The Offer will remain open for acceptance during the period commencing on 13 October 2020, and ending at 7.00pm Sydney time on 13 November 2020, unless it is withdrawn or extended under the Corporations Act.

6. How to accept this Offer

All of your holding

6.1 This Offer is for all of your CIE Shares.

Acceptance procedure for CIE Shareholders

6.2 The acceptance procedure will depend on whether your CIE Shares are in a CHESS Holding or an Issuer Sponsored Holding.

If you hold your CIE Shares in a CHESS Holding

- 6.3 You must comply with the ASX Settlement Operating Rules.
- 6.4 If you are a sponsored CHESS holder of CIE Shares, to accept the Offer you must:
 - (a) instruct your Controlling Participant to initiate acceptance of the Offer on your behalf under Rule 14.14 of the ASX Settlement Operating Rules;
 - (b) complete, sign and send the accompanying Acceptance Form directly to your stockbroker or Controlling Participant; or
 - (c) complete and sign the Acceptance Form and mail or deliver it to the address below and authorise WAM to instruct your Controlling Participant to initiate acceptance of the Offer on your behalf,

before the end of the Offer Period.

If you hold your CIE Shares in an Issuer Sponsored Holding

6.5 If your CIE Shares are in an Issuer Sponsored Holding, then to accept this Offer you must complete, sign and return the enclosed Acceptance Form in accordance with the instructions contained within it and deliver it or send it by post so that it is received by WAM's share registry of the Offer before the end of the Offer Period.

If you are a Broker or a non-Broker participant

6.6 If you are a Broker or non-Broker participant, initiate acceptance under the requirements of the ASX Settlement Operating Rules before the end of the Offer Period.

Mailing and delivery addresses

6.7 The mailing and delivery address for completed Acceptance Forms and any associated documents are as follows:

By mail:

Contango Income Generator Limited Takeover C/- Boardroom Pty Limited GPO Box 3993 Sydney NSW 2001

Hand delivered:

Boardroom Pty Limited Level 12, 225 George Street Sydney NSW 2000

6.8 A reply-paid envelope (not able to be used by CIE Shareholders outside Australia) is enclosed for your convenience.

Acceptance Form

6.9 The Acceptance Form which accompanies this Offer forms part of this Offer. The requirements on the Acceptance Form must be observed in accepting this Offer for your CIE Shares.

Power of attorney, deceased estate

- 6.10 When accepting this Offer, you must also forward for inspection:
 - (a) if the Acceptance Form is signed by an attorney, the power of attorney; and
 - (b) if the Acceptance Form is signed by the executor of a will or the administrator of the estate of a deceased CIE Shareholder, the relevant grant of probate or letters of administration.

When acceptance is complete

- 6.11 Acceptance of this Offer will not be complete until the completed Acceptance Form has been received at the address set out in clause 6.7 of Schedule 1 and the requirements of this clause have been met, provided that:
 - (a) WAM may in its sole discretion waive any or all of those requirements at any time; and
 - (b) where those requirements have been complied with for some but not all of your CIE Shares, WAM may, in its sole discretion, deem your acceptance of

this Offer complete in respect of those CIE Shares for which the requirements have been complied with but not in respect of the remainder.

6.12 If your Acceptance Form is returned by post it will be taken as received before the end of the Offer Period if the envelope in which it is sent is postmarked before the end of the Offer Period.

7. Provision of consideration

When consideration is provided

- 7.1 Subject to clauses 7.3 and 7.4 of this Schedule 1, you will receive the consideration you are entitled to within one month of the later of:
 - (a) the date you accept the Offer; and
 - (b) the date the Offer becomes unconditional.
- 7.2 In any event, you will receive the consideration you are entitled to within 21 days after the end of the Offer Period (assuming all Defeating Conditions are satisfied or waived).

Payments to Foreign Shareholders

- 7.3 If you accept the Offer and you are a Foreign Shareholder and WAM does not make a determination under Section 11.12 of this Bidder's Statement:
 - (a) WAM will arrange for the issue to the Nominee of the number of WAM Shares to which you and all other Foreign Shareholders would have been entitled under the Offer but for clause 3.1 of this Schedule 1;
 - (b) WAM will cause the Nominee to offer for sale the WAM Shares as soon as practicable, or otherwise in the manner, at the price and on terms as are determined by the Nominee;
 - (c) WAM will cause the Nominee to pay to you the amount calculated under the following formula:

Net Proceeds of Sale x Your WAM Shares

Total Nominee WAM Shares

where:

- (i) Net Proceeds of Sale means the amount which is received by the Nominee upon the sale of all WAM Shares under this clause 7.3 less brokerage and other sale expenses;
- (ii) Your WAM Shares means the number of WAM Shares which would, but for clause 3.1 of this Schedule 1, have been allotted to you under the Offer:

- (iii) Total Nominee WAM Shares means the total number of WAM Shares allotted to the Nominee under this clause 7.3;
- (d) you will receive your share of the proceeds of this sale in Australian currency;
- (e) payment will be made by cheque posted to you at your risk by ordinary mail to the address provided on your Acceptance Form; and
- (f) under no circumstances will interest be paid on your share of the proceeds of this sale, regardless of any delay in remitting these proceeds to you.

Where additional documents are required

- 7.4 Where the Acceptance Form requires additional documents to be given with your acceptance (such as a power of attorney), if the documents are given:
 - (a) with your acceptance, WAM will provide the consideration in accordance with clause 7.1 of this Schedule 1;
 - (b) after acceptance and before the end of the Offer Period and the Offer is subject to a Defeating Condition at the time that WAM is given the documents, WAM will provide the consideration by the end of whichever period ends earlier:
 - (i) within one month after the contract resulting from your acceptance of the Offer becomes unconditional; and
 - (ii) 21 days after the end of the Offer Period;
 - (c) after acceptance and before the end of the Offer Period and the Offer is unconditional at the time that WAM is given the documents, WAM will provide the consideration by the end of whichever period ends earlier:
 - (i) one month after WAM is given the document; and
 - (ii) 21 days after the end of the Offer Period; or
 - (d) after the end of the Offer Period, WAM will provide the consideration within 21 days after the documents are given, but if at the time WAM is given the documents, the contract resulting from your acceptance of the Offer is still subject to one or more of the Defeating Conditions, WAM will provide the consideration within 21 days after that contract becomes unconditional.
- 7.5 If you do not give WAM the required additional documents within one month after the end of the Offer Period, WAM may, in its sole discretion, rescind the contract resulting from your acceptance of the Offer.

Where WAM is entitled to any Rights

7.6 If WAM becomes entitled to any Rights as a result of your acceptance of the Offer, it may require you to give WAM all documents necessary to vest those Rights in WAM

or otherwise give the benefit or value of those Rights. If you do not give those documents to WAM, or if you have received the benefit of those Rights, then WAM may reduce the number of WAM Shares to which you are otherwise entitled under this Offer by the amount equal to the value, (as reasonably assessed by WAM), of those Rights in accordance with clause 7.7 of this Offer.

7.7 If:

- (a) you have (or any previous holder of your CIE Shares has) received the benefit of any Rights (whether in respect of non-cash benefits or otherwise); or
- (b) you are (or any previous holder of your CIE Shares is) entitled to receive the benefit of any Rights under the terms that provide for or otherwise apply to those Rights (for example, if the Right is to receive a dividend, if you are (or any previous holder of your CIE Shares is) the registered holder of the CIE Share at the specified time for determining those entitled to the dividend; or
- (c) your CIE Shares were issued (or otherwise came into existence) on or after the record date in respect of any Rights to CIE Shareholders,

then WAM may reduce the number of WAM Shares to be issued to you under this Offer by the value (as reasonably assessed by WAM) of such Rights. If WAM does not or cannot, make such a reduction, you must pay such value or amount to WAM.

Payment to be mailed by cheque

7.8 Payment of any cash amount to which you are entitled (by virtue of you being a Foreign Shareholder) will be mailed by cheque (or otherwise as agreed by WAM) in Australian currency. Cheques will be sent to the address on the Acceptance Form by pre-paid ordinary mail or, if you have an overseas address, by pre-paid airmail.

Clearances for offshore residents

7.9 If, at the time of acceptance of this Offer, you are resident in or of a place outside Australia, you will not be entitled to receive any consideration under this Offer until all requisite authorities or clearances of the Reserve Bank of Australia (whether under the Banking (Foreign Exchange) Regulations 1959 (Cth) or otherwise), or of the Australian Taxation Office, have been obtained.

8. Conditions of the Offer

Conditions

- 8.1 This Offer and the contract that results from acceptance of this Offer are subject to fulfilment of the following Defeating Conditions:
 - (a) (Minimum acceptance) Before the end of the Offer Period, WAM has a Relevant Interest in at least 50.1% (by number) of CIE Shares;
 - (b) (No proposal to change or re-assign the Investment Management Agreement to a party other than Wilson Asset Management) Between the date of this Bidder's Statement and the end of the Offer Period, the

Investment Management Agreement is not varied or assigned and no new investment management agreement is entered into (or CIE enters into an agreement in relation to any of the foregoing) other than an assignment to, or entry into an agreement with, Wilson Asset Management;

- (c) (No Prescribed Occurrences) Between the date of this Bidder's Statement and the end of the Offer Period, none of the following occur (each, a Prescribed Occurrence):
 - (i) CIE converts all or any of its shares into a larger or smaller number of shares;
 - (ii) CIE or a Subsidiary of CIE resolves to reduce its share capital in any way;
 - (iii) CIE or a Subsidiary:
 - (A) enters into a buy-back agreement; or
 - (B) resolves to approve the terms of a buy-back agreement under section 257C(1) or 257D(1) of the Corporations Act;
 - (iv) CIE or a Subsidiary of CIE issues shares, or grants an option over its shares, or agrees to make such an issue or grant such an option;
 - (v) CIE or a Subsidiary of CIE issues or agrees to issue convertible notes;
 - (vi) CIE or a Subsidiary of CIE disposes or agrees to dispose, of the whole, or a substantial part of its business or property;
 - (vii) CIE or a Subsidiary of CIE grants, or agrees to grant, a security interest in the whole, or a substantial part, of its business or property;
 - (viii) CIE or a Subsidiary of CIE resolves to be wound up;
 - (ix) a liquidator or provisional liquidator of CIE or a Subsidiary of CIE is appointed;
 - (x) a court makes an order for the winding up of CIE or a Subsidiary of CIE;
 - (xi) an administrator of CIE or a Subsidiary of CIE is appointed under section 436A, 436B or 436C of the Corporations Act;
 - (xii) CIE or a Subsidiary of CIE executes a deed of company arrangement; or
 - (xiii) a receiver, or a receiver and manager, is appointed in relation to the whole, or a substantial part, of the property of CIE or a Subsidiary of CIE.

- (d) (No fall in NTA) the pre-tax NTA of CIE not declining by 7.5% or more below the pre-tax net NTA of CIE of \$0.746 per share, announced to the ASX in the Monthly NTA Statement 31 August 2020 on 4 September 2020;
- (e) (No market fall) the Dow Jones Industrial Average, in AUD terms, not closing at a level that is 5% or more below the level of that index at 5.00pm (Sydney time) on the trading day immediately prior to the Announcement Date and remaining at or below that level for at least two consecutive trading days;
- (f) (No material adverse change) Between the date of this Bidder's Statement and the end of the Offer Period, there is no occurrence of any change which has or could reasonably be expected to have a material adverse effect on the assets, liabilities, financial position, performance, profitability or prospects of CIE and/or its Subsidiaries;
- (g) (No material change to the business of CIE and/or its Subsidiaries) Between the date of this Bidder's Statement and the end of the Offer Period, none of the following occur, without the written consent of WAM:
 - (i) CIE or a Subsidiary of CIE, adopts a new constitution or makes any change to its constitution or passes any special resolution or proposes to do so;
 - (ii) CIE or a Subsidiary of CIE, gives or agrees to give any encumbrance (including a Security Interest or mortgage) over any of its assets otherwise than in the ordinary course of business; or
 - (iii) CIE or a Subsidiary of CIE, appoints any additional Director to its Board of Directors whether to fill a casual vacancy or otherwise.
- (h) (No regulatory action) Between the date of this Bidder's Statement and the end of the Offer Period:
 - (i) no preliminary or final decision, order or decree is made or issued;
 - (ii) no action, proceeding or investigation is announced, commenced or threatened; and
 - (iii) no application is made (other than by WAM or its Associates),

by or to any regulatory organisation or Governmental Agency which is likely to or threatens to restrain, prohibit, impede or otherwise materially adversely affect the making of the Offer, the acquisition of the CIE Shares by WAM, the rights of WAM in respect of CIE and the CIE Shares or the continued operation of the businesses of CIE or its Subsidiaries (other than any action or decision by or application to ASIC or the Takeovers Panel in exercise of the powers or discretions conferred by the Corporations Act).

Nature of conditions

- 8.2 Each of the Defeating Conditions:
 - (a) constitutes and will be construed as a separate, several and distinct condition;
 - (b) is a condition subsequent; and
 - (c) until the expiration of the Offer Period will be for the benefit of WAM alone and may be relied upon only by WAM.

Conditions apply to multiple events

- 8.3 Where an event occurs that would mean at the time the event occurs a Defeating Condition would not be satisfied, each Defeating Condition affected by that event becomes two separate conditions on identical terms except that:
 - (a) one of them relates solely to that event; and
 - (b) the other specifically excludes that event.
- 8.4 WAM may declare the Offer free, under clause 8.5, from either of those conditions without declaring it free from the other and may do so at different times. This clause may apply any number of times to a Defeating Condition (including a condition arising from a previous operation of this clause).

Effect of breach or non-fulfilment

- 8.5 The breach or non-fulfilment of any of the Defeating Conditions does not prevent a contract arising to acquire your CIE Shares resulting from your acceptance of this Offer. If at the end of the Offer Period, in respect of any Defeating Condition:
 - (a) WAM has not declared the Offers (and they have not become) free from that Defeating Condition; or
 - (b) that Defeating Condition has not been satisfied,

all contracts resulting from the acceptance of the Offers and all acceptances that have not resulted in binding contracts are void. In that case, WAM will return the Acceptance Form together with all documents to the address shown in the Acceptance Form.

WAM may decide the Offer is free from all or any of the Defeating Conditions

8.6 WAM may at any time at its sole discretion but in compliance with section 650F and section 630(1) of the Corporations Act, declare the Offer free from all or any of the Defeating Conditions by written notice to CIE not later than 6 November 2020.

Date for giving notice on status of Defeating Conditions

8.7 The date for giving a notice on the status of the Defeating Conditions as required by section 630(1) of the Corporations Act is the date seven days before the end of the

Offer Period, subject to variation under section 630(2) of the Corporations Act if the Offer Period is extended.

Effect of acceptance

- 8.8 Once you have accepted this Offer, you will be unable to revoke your acceptance and the contract resulting from your acceptance will be binding on you, unless:
 - (a) by the end of the Offer Period, the Defeating Conditions have not all been satisfied or waived in which case, this Offer will automatically terminate; and
 - (b) the circumstances set out in section 650E of the Corporations Act apply.
- 8.9 By signing and returning the Acceptance Form, you will be deemed to have:
 - (a) irrevocably accepted this Offer (and each variation of this Offer) in respect of all your CIE Shares despite any difference between that number and the number of CIE Shares shown in the Acceptance Form;
 - (b) irrevocably authorised WAM to alter the Acceptance Form on your behalf by:
 - (i) inserting correct details of your CIE Shares;
 - (ii) filling in any blanks remaining on the Acceptance Form; and
 - (iii) rectifying any errors in, and omissions from, the Acceptance Form,

as may be necessary to make the Acceptance Form a valid acceptance of this Offer and to enable registration of the transfer of your CIE Shares to WAM;

- (c) represented and warranted to WAM, as a condition of the contract resulting from your acceptance of the Offer, that at the time of acceptance and at the time of transfer to WAM:
 - (i) you have paid to CIE all amounts which are due for payment for your CIE Shares;
 - (ii) all of your CIE Shares are fully paid and free from all mortgages, charges, liens and other encumbrances of any nature; and
 - (iii) you have full power and capacity to sell and transfer those CIE Shares;
- (d) granted to WAM and each of its Directors power of attorney to sign any document or do anything on your behalf necessary to cause CIE to issue a replacement share certificate for those CIE Shares;
- (e) agreed to:
 - (i) do anything necessary to ratify any act taken or purported to be taken by WAM or any of its Directors under the power of attorney granted by clause 8.9(d) of this Schedule 1; and
 - (ii) indemnify WAM and each of its Directors in respect of that act;
- (f) from the time of acceptance until the Offer has become unconditional, until the earlier of the withdrawal of your acceptance under section 650E of the Corporations Act or the end of the Offer Period, irrevocably appointed WAM or any nominee of WAM as your attorney to exercise all your powers and rights attaching to your CIE Shares including (without limitation) to requisition, convene, attend and vote at all general meetings of CIE or appoint a proxy or proxies to attend and vote on your behalf at that meeting, where the passing or defeat of any resolution before the meeting might adversely affect the likelihood of the Offer being successful and to do so in a way which would prevent that effect arising;
- (g) on this Offer or any takeover contract becoming unconditional, agreed not to attend or vote in person at any general meeting of CIE, and irrevocably appointed WAM or any nominee of WAM as your attorney to exercise all your powers and rights attaching to your CIE Shares including (without limitation):
 - (i) to requisition, convene, attend and vote at all general meetings of CIE or appoint a proxy or proxies to attend and vote on your behalf at that meeting; and
 - (ii) sign all documents and resolutions relating to your CIE Shares, and generally to exercise all powers and rights which you may have as a CIE

Shareholder and perform actions as may be appropriate to vest good title in your CIE Shares in WAM;

- (h) agreed that in exercising the powers conferred by the power of attorney in clause 8.9(f) of this Schedule 1, WAM is entitled to act in the interests of WAM;
- (i) agreed not to exercise or purport to exercise any of the powers conferred on WAM or its nominee in clause 8.9(f) of this Schedule 1;
- (j) agreed to accept the WAM Shares, to which you have become entitled by acceptance of this Offer, subject to the constitution of WAM and have authorised WAM to place your name on the register of members for those WAM Shares; and
- (k) if your CIE Shares are in a CHESS Holding and you sign and return the Acceptance Form, irrevocably authorised WAM:
 - to instruct your Controlling Participant to initiate acceptance of this Offer for your CIE Shares under the ASX Settlement Operating Rules;
 and
 - (ii) to give any other instructions in relation to your CIE Shares to your Controlling Participant on your behalf under the sponsorship agreement between you and that Controlling Participant.

Withdrawal

8.10 WAM may withdraw unaccepted Offers at any time with the written consent of ASIC and subject to the conditions (if any) specified in that consent.

Variation

- 8.11 WAM may vary this Offer in accordance with the Corporations Act.
- 9. Acceptances by transferees and nominees

Who may accept the Offer

- 9.1 During the Offer Period:
- (a) any person who is able to give good title to a parcel of CIE Shares may accept (if they have not already accepted an offer in the form of this Offer) as if an offer on terms identical with this Offer has been made to that person; and
- (b) any person who holds one or more parcels of CIE Shares as trustee, nominee, or otherwise on account of another person, may accept as if a separate offer had been made in relation to:
 - (iii) each of those parcels; and
 - (iv) any parcel they hold in their own right.

Holding shares

9.2 A person is taken to hold CIE Shares if the person is, or is entitled to be registered as, the holder of those CIE Shares.

Holding shares on trust or as a nominee

- 9.3 A person is taken to hold CIE Shares on trust for, as nominee for, or on account of, another person if that person:
 - (a) is entitled to be registered as the holder of CIE Shares; and
 - (b) holds the interest in the CIE Shares on trust for, as nominee for, or on account of, that other person.

Effective acceptance

- 9.4 An acceptance of an Offer under clause 9.1(b) of this Schedule 1 is ineffective unless:
 - (a) the person who holds on account of another person, gives that other person written notice stating that the CIE Shares consist of a separate parcel; and
 - (b) the acceptance specifies the number of CIE Shares in that parcel.

Notice of acceptance

- 9.5 A notice under clause (a) of this Schedule 1 must be made in writing.
- 9.6 A person may, at the one time, accept for two or more parcels under clause 9.1(b) of this Schedule 1 as if there had been a single offer for a separate parcel consisting of those parcels.

10. Other matters

Notices and other communications

- 10.1 Subject to the Corporations Act, a notice or other communication given by WAM to you in connection with this Offer is deemed to be duly given if it is in writing and:
 - (i) is delivered at your address as recorded on the register of members of CIE or the address provided in the Acceptance Form; and
 - (ii) is sent by pre-paid ordinary mail, or in the case of an address outside Australia by pre-paid airmail, to you at either of those addresses.

Return of documents

- 10.2 If:
 - (iii) this Offer is withdrawn after your Acceptance Form has been sent to WAM, but before it has been received; or
 - (iv) for any other reason WAM does not acquire the CIE Shares to which your Acceptance Form relates,

WAM will despatch at your risk your Acceptance Form together with all other documents forwarded by you.

11. Costs and expenses

- 11.1 All costs and expenses of the preparation, despatch and circulation of the Offers will be paid by WAM. There should be no stamp duty payable by CIE Shareholders upon the issue to them of WAM Shares. Should any stamp duty be payable on the transfer of CIE Shares to WAM pursuant to the Offer due to a change in the relevant legislation during the Offer period, these costs will be paid by WAM.
- 11.2 If your CIE Shares are held in a CHESS Holding or you hold your CIE Shares through a bank, custodian or other nominee, you should ask your Controlling Participant (usually your Broker or the bank, custodian or other nominee) whether it will charge any transaction fees or service charges connected with you accepting the Offer. These costs will not be paid by WAM.
- 11.3 If you are a Foreign Shareholder, the cash proceeds that you will be sent (following the sale of WAM Shares that you would otherwise have been entitled to receive under the Offer) net of transaction costs, will be paid by cheque in Australian dollars.

12. Official Quotation of WAM Shares

- 12.1 Subject to clause 1 of this Schedule 1, the WAM Shares offered by WAM as consideration under the Offer will rank equally with WAM Shares on issue at the Register Date.
- 12.2 WAM has been admitted to the Official List of ASX and shares of the same class as those to be issued as consideration under the Offer have been granted Official Quotation by ASX.
- 12.3 An application will be made within seven days after the start of the bid period (that is, within seven days after the date of this Bidder's Statement) to the ASX for Official Quotation of the WAM Shares issued pursuant to the Offer. However, Official Quotation is not granted automatically on application.
- 12.4 Under section 625(3) of the Corporations Act, the Offer is subject to a condition that application for admission to quotation of the WAM Shares issued under the Offer is made within seven days after the start of the bid period and permission for admission to quotation will be granted no later than seven days after the end of the bid period. This condition may not be waived.

Schedule 2 – WAM Announcements since 30 June 2020

Date	Announcement
23/09/2020	CLF: Target's Statement
22/09/2020	Change in substantial holding for KBC
21/09/2020	Proposed issue of Securities - WAM
21/09/2020	Change in substantial holding for CLF
18/09/2020	Bidder's Statement for CLF T/O Bid - offer now open
18/09/2020	Change in substantial holding for MEL - Correction
18/09/2020	Change in substantial holding for MEL
16/09/2020	Change in substantial holding for WCG
16/09/2020	Ceasing to be a substantial holder for CGL
11/09/2020	August 2020 Investment Update
8/09/2020	Appendix 3X Initial Director's Interest Notice - M Pancino
4/09/2020	WAM Capital, WAM Research, WAM Active Investor Q&A
4/09/2020	WAM Capital bids for Concentrated Leaders Fund
4/09/2020	CLF: Off Market Takeover Bid-Take No Action
3/09/2020	Off-Market Takeover Bid for Concentrated Leaders Fund (CLF)
01/09/2020	WAM Capital appoints Director Matthew Pancino
28/08/2020	Change in substantial holding for VVA
28/08/2020	Change in substantial holding for KBC
25/08/2020	Change in substantial holding for ADH
25/08/2020	2020 Full Year Results
25/08/2020	Appendix 4E and Annual Report
25/08/2020	Corporate Governance Statement
25/08/2020	Appendix 4G
21/08/2020	Change in substantial holding for TRS
18/08/2020	CIE proposal for new manager and investment mandate
17/08/2020	Ceasing to be a substantial holder for BWX
14/08/2020	July 2020 Investment Update
13/08/2020	Becoming a substantial holder for WCG
13/08/2020	Change in substantial holding for YOW
4/08/2020	Change in substantial holding for KBC
3/08/2020	Change in substantial holding for ADH

Date	Announcement
3/08/2020	Becoming a substantial holder for TRS
29/07/2020	Change in substantial holding for KBC
24/07/2020	Change in substantial holding for KBC
14/07/2020	June 2020 Investment Update
8/07/2020	Change in substantial holding for BWX
8/07/2020	Investment portfolio outperformance, 15.5cps ff full yr div
8/07/2020	Dividend/Distribution - WAM



CEPTANCE AND TRANSFER FORM

in respect of the Offer by WAM Capital Limited ACN 086 587 395 (WAM) to acquire all of your Shares in Contango Income Generator Limited ACN 160 959 991 (CIE)

THIS IS AN IMPORTANT DOCUMENT

IF YOU DO NOT UNDERSTAND IT PLEASE CONSULT YOUR FINANCIAL, LEGAL OR OTHER PROFESS

1)

3)

IF 100 DO NOT GINDERSTAND IT FELASE CONSOLT TOOK FINANC	IAL, LEGAL OR OTTILK PROPESSIONAL	ADVISER IMMEDIATELT
STEP 1 – Check your details below		
	Number of CIE Shares you hold	
	HIN/SRN'	
		A new SRN may be allocated to you solely for the purpose of this Offer.
	Sub-register	
	Barco	ode to be printed here
Use this Acceptance Form to accept the Offer by WAM for all your CIE Shares on the or supplements to it. You should read the Bidder's Statement, which accompanies to same meaning as in the Bidder's Statement unless otherwise defined. By accepting (even if different to the number of shares stated above). You cannot accept for only If you need help completing this Acceptance Form or have any enquiring in relation	his Acceptance Form. Capitalised terms u the Offer, your acceptance will be treate part of your CIE Shares. n to the Offer, please contact Boardroom	ised in this Acceptance Form have the d as being for ALL of your CIE Shares
within Australia) or +61 2 8023 5472 (callers outside Australia) Monday to Friday be	·	
STEP 2 – For Issuer Sponsored Holdings only	OFFER CON	ISIDERATION
If your CIE Shares are held on the Issuer Sponsored sub-register or if at the time of your acceptance you are entitled to be (but are not yet) registered as a holder of your CIE Shares, to accept the Offer you must sign below and return this form .	et) registered as 1 WAM Share for every 3 of your CIE Shares	
STEP 3 – For CHESS Holdings only		
If your CIE Shares are held on the CHESS sub-register, to accept the Offer you can	BROKER DETAILS	(CHESS HOLDINGS ONLY)
either:		
Contact your Controlling Participant – normally your Broker – and instruct them to accept the Offer on your behalf (If you do that and your Controlling Participant follows those instructions, you do not need to complete and return	Your Broker's name	
this Acceptance Form); OR	Your Broker's address	
 If you want WAM to contact your Controlling Participant on your behalf, write their details here and sign and return this form. By doing so you authorise WAM to instruct your Controlling Participant to initiate acceptance of the Offer and to take all other steps necessary to cause acceptance of the Offer in accordance with the ASX Settlement Operating Rules. 	Your Broker's telephone number	
STEP 4 – Sign as indicated below		
Ne, the shareholder(s) named above, being the holder(s) of CIE Shares:		
ACCEPT the Offer in respect of ALL of my/our CIE Shares;		
AGREE TO TRANSFER my/our CIE Shares to WAM for the consideration specifi	ed in the Offer, and in accordance with the	e terms of the Offer;
AUTHORISE WAM, its officers and agents to correct any errors in or omissions Offer and enable registration of the transfer of the CIE Shares to WAM; and	s from this Acceptance Form to make it a	n effective acceptance of the
AGREE to be bound by the terms of the Offer.		
Please refer overleaf for further instruc	ctions on how to complete this for	m
this form is signed under power of attorney, the attorney declares that they have no		
Individual or Shareholder 1 Share	holder 2	Shareholder 3

Sole Director and Company Secretary Director/Company Secretary Director Telephone number (business hours): Date: / /2020

Your acceptance must be received BEFORE the Offer Period ends (being 7:00pm (Sydney time) on 13 November 2020 unless extended). If the Acceptance Form is sent by mail, you may use the enclosed reply-paid envelope.

How to complete this Acceptance Form and ACCEPT the Offer

Your name and address

Your pre-printed name and address is as it appears on the share register of CIE. If you are Issuer Sponsored and this information is incorrect, please make the correction on this form and initial the correction. Shareholders sponsored by a Broker on the CHESS sub-register should advise their Broker of any changes.

2. Issuer Sponsored Holdings (as indicated by "Issuer Sponsored" appearing next to "Sub-register" on this Acceptance Form)

If your CIE Shares are in an Issuer Sponsored Holding, or if you are not yet registered as the holder of your CIE Shares, then to accept the Offer, you must fill out this Acceptance Form overleaf and return it to one of the addresses shown below.

3. CHESS Holdings (as indicated by "CHESS" appearing next to "Sub-register" on this Acceptance Form)

If your CIE Shares are in a CHESS Holding, you do not need to complete and return this Acceptance Form to accept the Offer. You can contact your Controlling Participant, normally your Broker, and instruct them to accept the Offer on your behalf. If you decide to use this Acceptance Form, follow the instructions below.

It is the responsibility of the eligible shareholder to allow sufficient time for their Controlling Participant to initiate acceptance on their behalf in accordance with ASX Settlement Operating Rule 14.14. You must ensure that this Acceptance Form is received in sufficient time before the end of the Offer Period to enable us to instruct your Controlling Participant to effect acceptance on CHESS during business hours.

If your holding is CHESS sponsored and you send your Acceptance Form to Boardroom Pty Limited, they will send the relevant acceptance message to CHESS for forwarding to your Controlling Participant for acknowledgement. WAM and Boardroom Pty Limited will not be responsible for any delays incurred by this process.

If you sign and return this Acceptance Form to one of the below addresses either in respect of an Issuer Sponsored Holding or so that contact may be made with your Controlling Participant on your behalf, you warrant to WAM (and authorise WAM to warrant on your behalf) that you have full legal and beneficial ownership of the CIE Shares and that WAM will acquire them free from all mortgages, charges, liens, encumbrances (whether legal or equitable), restrictions on transfer and free from any third party rights.

4. Signature(s)

You must sign the form as follows in the space provided overleaf:

Joint holding: Where the shareholding is in more than one name, all of the CIE Shareholders must sign.

Power of attorney: To sign under power of attorney, you must attach a certified copy of the power of attorney to this form when you return it. By signing this form under a power of attorney, you will have declared that you have no notice of revocation of the power and are able to further delegate power under it under the Bidder's Statement.

Deceased estate: All executors must sign and, if not already noted by the CIE registry, attach a certified copy of probate, letters of administration or grant accompanied (where required by law for the purpose of the transfer) by a certificate of payment of death or succession duties and (if necessary) a statement in terms of section 1071B(9)(b)(iii) of the Corporations Act.

Companies: This form must be signed by either 2 directors or a director and a company secretary. Alternatively, where the company has a sole director and, pursuant to the Corporations Act, there is no company secretary, or where the sole director is also the sole company secretary, that director may sign alone. Delete titles as applicable.

Additional notes:

- 1. Sold all your CIE Shares if you have sold all of your CIE Shares, please send this form and your Bidder's Statement to the stockbroker who acted on your behalf.
- 2. Recently bought or sold CIE Shares if you have recently bought or sold any CIE Shares, your holding may differ from that shown on the front of this form. If so, please alter the number of CIE Shares shown as your registered holding on the front of this form to the number of CIE Shares you now hold (including any CIE Shares of which you are entitled to become registered as holder), initial the alteration and indicate the name of the stockbroker who acted for you.

Information you supply on this Acceptance Form will be used by WAM and Boardroom Pty Limited for the primary purpose of processing your acceptance of the Offer and to provide you with the consideration payable under the Offer. This information may be disclosed to WAM's professional advisers, securities brokers, printing and mailing providers and other third parties in connection with the Offer. If you do not supply this information, your acceptance may not be processed and you may not receive the consideration payable. You may have rights to access the personal information you have supplied. Please see Boardroom Pty Limited's privacy policy on its website.

This Acceptance Form must be received at the relevant address shown below before the end of the Offer Period.

Postal address

Contango Income Generator Limited Takeover C/- Boardroom Pty Limited GPO BOX 3993 SYDNEY NSW 2001

Delivery in person

Boardroom Pty Limited Level 12, 225 George Street SYDNEY NSW 2000

If the Acceptance Form is sent by mail, you may also use the enclosed reply-paid envelope.

Neither WAM nor Boardroom Pty Limited takes any responsibility if you lodge this Acceptance Form at any other address or by any other means.